

March 2023

The Financial Macro-econometric Model (FMM, 2022 Version)

Financial System and Bank Examination Department, Bank of Japan ABE Nobuhiro CHIKAMATSU Kyosuke KANAI Kenji KAWASUMI Yusuke MUNAKATA Ko NAKAYAMA Koki OKUDA Tatsushi TAKANO Yutaro

Please contact below in advance to request permission when reproducing or copying the content of this paper for commercial purposes. Financial System and Bank Examination Department, Bank of Japan Tel: +81-3-3279-1111 Please credit the source when reproducing or copying the content of this paper. March 2023 Financial System and Bank Examination Department, Bank of Japan ABE Nobuhiro¹ CHIKAMATSU Kyosuke² KANAI Kenji³ KAWASUMI Yusuke⁴ MUNAKATA Ko⁵ NAKAYAMA Koki⁶ OKUDA Tatsushi⁷ TAKANO Yutaro⁸

The Financial Macro-econometric Model (FMM, 2022 Version)⁹

Abstract

The Financial Macro-econometric Model (FMM) is the model that the Bank of Japan (BOJ) employs in its macro stress testing to examine the risk resilience of Japan's financial system in a comprehensive and quantitative manner. The BOJ semiannually publishes the results of its analyses based on this model in the *Financial System Report*. The FMM is also used in the simultaneous stress testing based on common scenarios conducted periodically with the Financial Services Agency of Japan. Key characteristics of the FMM are that it (1) explicitly captures feedback loops between the domestic banking sector and the real economy, and (2) makes it possible to calculate the variables of interest (e.g. amount of loans and capital adequacy ratios of Japanese banks), not only at the sector level but also at the individual bank level. Since its development in 2011, the FMM has been continuously improved to reflect new developments in economic and financial conditions and to better incorporate the transmission mechanisms of financial shocks into the macro stress testing. This paper provides an outline of the basic macro stress testing framework and the FMM, and then describes the structure of the model as of September 2022 in detail.

¹ E-mail: nobuhiro.abe@boj.or.jp

² Currently at Research and Statistics Department; E-mail: kyousuke.chikamatsu@boj.or.jp

³ Currently at Personnel and Corporate Affairs Department; E-mail: kenji.kanai@boj.or.jp

⁴ E-mail: yuusuke.kawasumi@boj.or.jp

⁵ E-mail: kou.munakata@boj.or.jp

⁶ E-mail: kouki.nakayama@boj.or.jp

⁷ Currently at International Monetary Fund; E-mail: TOkuda@imf.org

⁸ E-mail: yuutarou.takano@boj.or.jp

⁹ The authors would like to thank AOKI Kosuke, INATSUGU Haruhiko, MIYAKAWA Daisuke, SUDO Nao, SUZUKI Koichiro, TAMANYU Yoichiro, and TOMURA Hajime for their helpful comments. In addition to the authors, ARAI Katsumi, KATAGIRI Mitsuru and MIURA Ko contributed considerably to the recent development of the FMM. The authors are also grateful to the external experts, who belong to international institutions, domestic and foreign public sectors, research institutions and financial institutions, for their insightful discussions. Any errors or omissions are the responsibility of the authors. The views expressed here are those of the authors and should not be ascribed to the Bank of Japan.

1. Introduction
2. Overview of Stress Testing
2.1. The macro stress testing framework
2.2. Overview of the FMM
2.3. Estimation method12
3. Structure of the Model of the Domestic Banking Sector
3.1. The investment and funding module15
3.2. The periodic profit/loss module19
3.2.1. PPNR excluding trading income21
3.2.2. Realized gains/losses on securities
3.2.3. Credit costs
3.3. Module for valuation gains/losses on securities holdings
3.4. Risk-weighted assets module
3.5. Capital adequacy module
4. Example of Macro Stress Testing
4.1. Scenario
4.2. Simulation results
5. Conclusion
Reference
Appendix 1: Structure of the Model of the Domestic Economy Sector
Appendix 2: Structure of the Model of the Domestic and Foreign Corporate Sectors
Appendix 3: Variables of Domestic and Foreign Financial Markets77
Appendix 4: Simulation Results of "1/2 Financial Stress Scenario"79
Appendix 5: Simulation under the Pandemic83
Appendix 6: Stress Testing Models Developed by the Central Banks in America and Europe

Outline

1. Introduction

The Bank of Japan (BOJ) regularly conducts macro stress testing to examine the stability of Japan's financial system and releases the results of its analysis twice a year in the *Financial System Report* (FSR). Macro stress testing is one analytical approach used to quantitatively examine the risk resilience of the financial system and is conducted based on stress scenarios assuming "severe but plausible" macroeconomic shocks to examine how the financial system would be affected. Specifically, the stress testing assumes the materialization of tail risks regarding the economic and financial environment (tail events) that, while not very likely, would cause extremely large stress in the financial system if they did occur, and forecasts banks' losses, capital adequacy ratios, etc., through simulations. Based on the results of these simulations, the FSR identifies the nature of risks faced by Japanese banks and assesses the risk resilience of the financial system as a whole. The results are also used to deepen the BOJ's dialogue with relevant parties in the world of finance at home and abroad.

Stress testing, which examines the resilience of financial institutions and the financial system overall based on a scenario assuming the materialization of the tail risks, has developed since the 1990s. Originally it was mainly used for risk management at individual financial institutions.¹ However, since the Federal Reserve Board (FRB) implemented the Supervisory Capital Assessment Program (SCAP) for 19 large U.S. financial institutions immediately after the global financial crisis in the late 2000s, stress testing has been widely used by national authorities and international organizations from the perspective of both microprudential and macroprudential policy.²

Stress testing is utilized by national authorities for many purposes. The FRB, the Bank of England (BOE), and the European Banking Authority (EBA)/European Central Bank (ECB) formulate stress scenarios and then conduct supervisory simultaneous stress tests of financial institutions in their jurisdiction based on a common stress scenario. They then use the results in supervisory policies for financial institutions, such as the approval

¹ Stress testing conducted by individual financial institutions aims to help financial institutions understand the risk characteristics of their own portfolios and is used by the board of directors and senior management to discuss those risks. As such, the stress testing conducted by financial institutions differs from that conducted by supervisors and central banks, which takes macroeconomic feedback effects into account. An overview of the risk management framework and tools used by financial institutions in the 1990s, such as the contents of stress tests, is provided by Fender, Gibson, and Mosser (2001) based on a survey by the Committee on the Global Financial System.

² Microprudential policy focuses on ensuring the soundness of individual financial institutions, while macroprudential policy focuses on ensuring the stability of the financial system by analyzing and assessing risks to the system and designing rules and policies based on these analyses and assessments. For more details on the BOJ's views on macroprudential policy, see, for example, Bank of Japan (2011).

process for financial institutions' capital planning.³ Moreover, from a macroprudential perspective to monitor the financial system in the euro area, the ECB regularly conducts top-down stress testing.⁴ The top-down stress testing incorporates feedback loops between the financial sector and the real economy, and the results of the tests are published in its Financial Stability Review (ECB 2021a). As mentioned above, the BOJ also regularly publishes the results of its top-down stress testing in the FSR. In addition, with the aim of establishing comprehensive evaluation targets regarding the financial soundness of banks and promoting the development of risk management arrangements through dialogue with banks, the BOJ, jointly with the Financial Services Agency (FSA), has been conducting "simultaneous stress testing based on common scenarios" focusing on major banks since 2020 (Bank of Japan and Financial Services Agency 2020).

Reflecting the expanding scope of the use of stress testing, authorities in major economies have been developing and improving models used for stress testing.⁵ In 2011, the BOJ developed the Financial Macro-econometric Model (FMM), a macroeconomic model that explicitly models feedback loops between the banking sector and the real economy, including the effects of the real economy on banks' capital adequacy ratios and the effects of financial intermediation activities on the real economy (Ishikawa et al. 2012). Since then, various improvements have been made in accordance with changes in the economic and financial environment and advances in academic research on both the theoretical and the empirical front. For example, from the perspective of refining the measurement of the transmission effects of tail risk, in 2012, the model was refined so that it could take into account the heterogeneity in the risk characteristics of individual banks' loan portfolios using granular financial data for individual banks and firms. Further, the model was improved to capture nonlinearities in the link between banks' capital adequacy ratios, profits, and the amount of loans as well as the impact of higher foreign currency funding

³ The FRB and the BOE use stress tests once a year and the EBA/ECB use them every other year for large financial institutions in their jurisdictions as part of the approval process for capital planning. For example, the FRB, through its Dodd-Frank Act Stress Tests and Comprehensive Capital Analysis and Reviews quantitatively examines financial institutions' capital adequacy and uses the results to determine the adequacy of their capital planning, including their dividends and stock buybacks. The BOE uses the results of its stress testing to set capital buffers for individual banks, to determine the appropriateness of dividends and stock buybacks, and to set countercyclical capital buffers to make provisions for macroprudential risks. The EBA/ECB use the stress test results to set supervisory capital buffer levels based on Pillar 2 and to determine the appropriateness of dividends and stock buybacks.

⁴ Baudino et al. (2018) classify stress testing conducted by financial institutions as bottom-up stress testing and stress testing using models developed by authorities as top-down stress testing.

 $^{^{5}}$ For a list of top-down stress testing models developed by central banks, see Appendix 6. For an international comparison of stress testing models, see Dent et al. (2016), Baudino et al. (2018), and Anderson et al. (2018).

costs when banks' own creditworthiness (capital adequacy ratio) declines. In addition, while initially only net interest income and domestic credit costs were modeled in banking sector, in 2014, realized and valuation gains/losses on securities holdings, credit costs on foreign lending, net non-interest income, and risk-weighted assets were also modeled in order to capture the wide variety of transmission channels through which tail risks affect banks. These changes are published in the form of research papers, which form the basis for discussions with related parties in the world of finance and researchers at universities and research institutions.⁶

Against the background, this paper explains the structure of the FMM as of September 2022. While previous papers explaining the FMM and macro stress testing, such as Kitamura et al. (2014), have focused on feedback loops between the banking sector and the real economy as well as specific topics such as interest rate risk and credit risk, this paper provides a comprehensive and detailed explanation of the structure of the model of the domestic banking sector, given that stresses in the real economy and financial markets can have a wide range of effects on the banking sector. The paper is organized as follows. Section 2 provides an outline of the FMM and the macro stress testing framework, while Section 3 explains the structure of the model of the domestic banking sector in the FMM. Section 4 describes the behavior of the FMM assuming a tail event in the form of a financial crisis, taking the "financial stress scenario" used in the macro stress testing in the April 2022 issue FSR as an example. Section 5 discusses future issues. Meanwhile, the appendix materials at the end of the paper provide the modeling of GDP demand items in the domestic economy of the FMM (Appendix 1), the modeling of the domestic and foreign corporate sectors (Appendix 2), a list of scenario variables for the domestic and foreign financial markets (Appendix 3), the results of simulation under a scenario in which the stress is half the size of that in the "financial stress scenario" (Appendix 4), the framework for the analysis of the stress imposed on the real economy by the outbreak of the Covid-19 pandemic, a tail event that did not take the form of a financial crisis (Appendix 5), and an overview of stress testing models used by central banks in America and Europe (Appendix 6).

⁶ For details on past improvements since the release of the first model in Ishikawa et al. (2012), see Kitamura et al. (2014) and Financial System Report - Annex, "The Financial Macro-econometric Model (FMM, March-2020 Version): Overview and Recent Developments" (published in August 2020). For analyses using the FMM, see Kamada and Kurachi (2012), Kawata et al. (2012), and Kawata et al. (2013).

2. Overview of Stress Testing

2.1. The macro stress testing framework

The BOJ's macro stress testing framework based on the FMM is as shown in Chart 1.

Chart 1. Macro stress testing framework



To start with, tail risk scenarios expressing the stress on the real economy and financial markets in the form of a deterioration in economic variables such as GDP and financial market variables such as stock prices are formulated (shown as (1) in the chart).⁷ In order to accurately identify the nature of risks to the financial system through macro stress testing, the scenarios need to reflect the financial and economic conditions at the time and the BOJ's perception of risks.^{8,9} Moreover, in certain scenarios, the BOJ may modify the

⁷ Looking at examples of macro stress tests conducted in various countries, in addition to stress scenarios, a baseline scenario in line with the forecasts of financial market participants and economists is often developed. This is thought to facilitate the identification of the nature of risks to the financial system by analyzing the differences between the stress scenarios and the baseline scenario. Hence, the BOJ's macro stress tests also include a baseline scenario in addition to stress scenarios. See Basel Committee on Banking Supervision (2009), Dees, Henry, and Martin (2017), and Liang (2018) for international discussions on the setting of stress test scenarios.

⁸ It should be noted that the stress scenarios in the BOJ's macro stress testing are designed to assess the stability of the financial system and the capital adequacy of individual banks under stress conditions; they are not the events that are assumed to be the most likely to occur. In other words, the stress scenarios are based on the conditions of the financial system, the risk profiles of individual banks, and other economic and financial conditions at the time, and are designed to effectively examine changes in economic and financial conditions that could pose a threat to the stability of Japan's financial system and the soundness of individual banks. Therefore, they are completely different in nature from economic forecasts that present scenarios with a high likelihood of coming to pass. They represent neither the BOJ's outlook for the future economic and financial environment, asset prices, policy conduct, nor the likelihood of the outcome.

⁹ In the United States, the approach for constructing stress scenarios and the degree of stress in the

FMM in order to accurately capture scenario-specific risks (represented by (2) in the chart).¹⁰ The economic and financial variables in the scenarios are then inserted into the FMM as input variables, and simulations are conducted for the future developments of the variables of individual banks such as their amount of loans and of macroeconomic variables such as GDP, taking into account the feedback effects ((3) in the chart).¹¹ Finally, the resilience of the financial system as a whole to tail risk events is assessed by examining the predicted profitability and capital adequacy ratios of all banks generated in the simulation ((4) in the chart).¹²

The primary interest in macro stress testing is the level of capital adequacy ratios during the simulation period, especially at the end of the simulation period. This level depends on three factors: the size and composition of banks' assets and liabilities and capital adequacy ratios at the end of the pre-simulation period, the severity of the stress in the scenario, and the structure of the FMM. The lower capital adequacy ratios at the end of the pre-simulation period are, the greater the severity of stress is, and the more sensitive

FRB's Dodd-Frank Act Stress Tests are set out in the "Policy Statement on the Scenario Design Framework for Stress Testing" (12 CFR 252, Appendix A), and scenarios are formulated in line with this (Board of Governors of the Federal Reserve System 2019). In Europe, the European Systemic Risk Board (ESRB), which is in charge of macroprudential issues, sets scenarios for the EBA/ECB's EU-wide stress tests in collaboration with the EBA/ECB (European Systemic Risk Board 2020). Moreover, in the IMF's Financial Sector Assessment Program stress tests, the degree of economic deterioration in case of a tail event assuming current financial conditions is calculated separately using Growth at Risk analysis (Adrian, Boyarchenko, and Giannone 2019, Adrian et al. 2022), and the scenario is set by calibrating financial and economic variables to reflect this deterioration (Adrian et al. 2020).

¹⁰ For example, in the macro stress testing in the October 2020 FSR, changes were made, among other things, to the credit cost model in order to incorporate the severe impact of the Covid-19 pandemic on economic activity in certain industries, such as face-to-face services, and the effect of policy measures to support corporate financing.

¹¹ In the scenario assuming a global financial crisis, in which there is a negative shock to foreign economies and financial markets that propagates to the domestic economy, the scenario for GDP uses endogenous values in the model. On the other hand, in scenarios where there is a direct negative shock to the domestic economy, such as the pandemic, exogenous values are used for the scenarios for GDP.

¹² In terms of assessment criterion for simulated capital adequacy ratios, the capital adequacy requirements of Basel III, which were introduced in 2010 (in particular, the minimum required Common Equity Tier 1 ratio of at least 4.5 percent) are used for internationally active banks. On the other hand, for banks that conduct business only in Japan and are not internationally active, the domestic capital adequacy requirements (of at least 4 percent) are used. In addition, among internationally active banks, the following levels of capital adequacy ratios are determined as loss absorbency requirements as of the writing of this paper; for global systemically important banks (G-SIBs), 8.5 percent for Mitsubishi UFJ Financial Group (consisting of a minimum CET1 ratio of 4.5 percent plus a capital conservation buffer of 2.5 percent and an additional buffer of 1.5 percent) and 8.0 percent for Mizuho Financial Group and Sumitomo Mitsui Financial Group (additional buffer: 1 percent); for domestic systemically important banks (D-SIBs), 7.5 percent for Sumitomo Mitsui Trust Holdings (additional buffer: 0.5 percent); and 7.0 percent for all others (no additional buffer).

the model is to stress, the lower capital adequacy ratios are during the simulation period. Meanwhile, it should be noted that while, in the FMM, banks' financial variables in the pre-simulation period are updated semiannually for banks excluding *shinkin* banks and annually for *shinkin* banks, during the simulation period, changes in financial variables are projected on a quarterly basis based on the quarterly scenario for economic and financial variables.

2.2. Overview of the FMM

As a stress testing model, the purpose of the FMM, as mentioned, is to quantitatively assess the risk resilience of Japan's financial system by simulating the amount of losses and capital adequacy ratios of Japanese banks in the event that tail risks to economic and financial conditions materialize.

The FMM is a macroeconomic model that broadly consists of two sectors: the banking sector and the real economy. For the banking sector, financial and other variables are modeled for each bank, yielding not only macroeconomic aggregate variables such as total loans, but also variables for individual banks. The model focuses on banks (major banks, regional banks, and *shinkin* banks) that hold current accounts at the BOJ.¹³ As a result of modeling the detailed account items for each of these banks to calculate their capital adequacy ratios in line with regulations, the number of variables included in the model is approximately 140,000 (of which 70,000 are endogenous variables and 70,000 are exogenous variables). The quantitative relationships among the variables, including the behavioral equations for banks, such as the effect of changes in GDP on the amount of lending, are estimated using panel estimation and other approaches to take heterogeneity across banks to some extent into account. In terms of the nature of the model, it can be regarded as a macroeconomic model belonging to the family of semistructural models, which impose some theoretical constraints on individual equations based on economic theory while selecting explanatory variables that fit the actual data well.¹⁴ In principle, the explanatory variables in the model are those for which the

¹³ Major banks comprise the following 10 banks: Mizuho Bank, MUFG Bank, Sumitomo Mitsui Banking Corporation, Resona Bank, Saitama Resona Bank, Mitsubishi UFJ Trust and Banking Corporation, Mizuho Trust and Banking Company, Sumitomo Mitsui Trust Bank, Shinsei Bank, and Aozora Bank. Regional banks comprise the 62 member banks of the Regional Banks Association of Japan (Regional banks I) and the 37 member banks of the Second Association of Regional Banks (Regional banks II). *Shinkin* banks are the 247 *shinkin* banks (as of the end of March 2022) that hold current accounts at the BOJ. Taken together, these banks account for about 80–90 percent of lending by private-sector financial institutions.

¹⁴ On the other hand, models built strictly based on relationships derived from economic theory, such as dynamic stochastic general equilibrium models, are called structural models.

estimated coefficients have the expected sign suggested by economic theory and are statistically significantly different from zero.

Relatedly, the BOJ maintains another semi-structural model for macroeconomic forecasting, the Quarterly-Japanese Economic Model (Q-JEM).¹⁵ In contrast to the Q-JEM, the FMM, which aims to calculate the capital adequacy ratios of individual banks as well as financial system as a whole, does not model real economic variables in detail, while balance sheet and income statement items of individual banks are modeled in detail, reflecting the purpose of the model.

To provide an overview of the structure of the FMM, it is useful to regard the model as consisting of four sectors: the financial sector, which is divided into (1) domestic and foreign financial markets sector and (2) the domestic banking sector,¹⁶ and the real economy, which is divided into (3) foreign economies sector and (4) the domestic economy sector (Chart 2).

The following subsections provide an overview of each sector and its relationship to the other sectors, starting with foreign economies, followed by domestic and foreign financial markets, the domestic economy, and the domestic banking sector.



Chart 2. The four sectors constituting the FMM

¹⁵ For an outline of Q-JEM, see Ichiue et al. (2009), Fukunaga et al. (2011), and Hirakata et al. (2019).

¹⁶ The foreign banking sector (such as U.S. and European financial institutions) is not modeled.

Foreign economies sector

Variables for foreign economies include real GDP, industry-level GDP, and financial variables of the corporate sector (the growth rate of sales and the interest coverage ratio, ICR¹⁷) for the United States, Europe (the euro area), and Asia and Pacific.¹⁸ Changes in the real GDP of the different regions affect Japan's domestic economy through changes in the demand for Japanese exports,¹⁹ while changes in the financial variables of the corporate sector in these regions change the foreign credit costs of banks in the domestic banking sector in proportion to their exposure. Including these variables means that spillover effects from abroad are incorporated into the model.

Domestic and foreign financial markets sector

To reflect the impact of developments in domestic and foreign financial markets, the model includes about 50 financial market variables covering stock markets, foreign exchange markets, bond markets, credit markets, and investment fund and alternative investment markets (see Appendix 3 for details). These variables capture the impact of changes in the prices of securities held by domestic banks on banks' realized and valuation gains/losses on securities as well as the direct impact of changes in Japanese stock prices on domestic households' consumption behavior through the wealth effect. In addition, an increase in domestic and foreign (U.S. and European) interest rates, for example, leads to a deterioration in the ICR of the corporate sector in domestic and foreign economies and hence increases domestic banks' credit costs in line with their exposure.

Domestic economy sector

Variables for the domestic economy primarily consist of macroeconomic variables such as nominal and real GDP and GDP demand components. For the corporate sector, in addition to business fixed investment as a GDP demand component, a range of financial variables are modeled (see Appendix 1 for details on the modeling of GDP demand components). Conversely, the government sector, including the central bank, is not explicitly modeled, and interest rates and government spending are exogenous variables. Inflation is in principle also treated as exogenous. As discussed below, these macroeconomic variables have an impact on the financial variables of domestic banks, such as their lending and credit costs.

¹⁷ The ICR is defined as (Operating profits + Interest and dividends received) / Interest payments.

¹⁸ Asia and Pacific consists of China, the NIES, the ASEAN countries, and Oceania.

¹⁹ For details, see the specification of the nominal and real export functions in Appendix 1.

Domestic banking sector

As outlined in Section 3, the domestic banking sector is modeled in detail, consisting mainly of variables for banks' assets and liabilities, such as investment and funding, periodic profits and losses, risk-weighted assets, and capital adequacy ratios. The model also incorporates various channels that affect the capital adequacy ratios of individual banks. Specifically, changes in financial variables relating to the corporate sector in the domestic and foreign economies, changes in banks' lending and funding rates, and changes in yields on, and market values of, securities holdings all affect the capital adequacy ratios through changes in their investment holdings and funding, periodic profits and losses, valuation gains/losses on securities holdings, risk-weighted assets, etc.

Importantly, linkages between the domestic economy and the domestic banking sector are explicitly incorporated.²⁰ In other words, a downturn in the domestic economy pushes down the amount of lending from the demand side by reducing the demand for funds in the lending market, and a deterioration in the financial conditions of the corporate sector leads to a reduction in lending from the supply side by lowering banks' capital adequacy ratios through higher credit costs. The reduction in lending due to supply-side factors further depresses the domestic economy by hampering investment and consumption.

It should be noted that with regard to the domestic banking sector, a distinction is made between banks excluding *shinkin* banks and *shinkin* banks, and that banks are further divided into internationally active and domestic banks,²¹ which are subject to different definitions and calculation methods with respect to their regulatory capital adequacy requirements, so that there are three types of banks in total. Similar to domestic banks,

²⁰ Although there were some studies highlighting the existence of linkages between the real economy and the financial sector (e.g., Bernanke, Gertler and Gilchrist 1996; Kiyotaki and Moore 1997), a widely accepted view among macroeconomists before the global financial crisis was that the financial sector was merely a mirror of the real economy. However, since the global financial crisis, the importance of frictions and shocks in the financial sector, including financial imbalances, for the real economy has been recognized, and there is a growing body of both theoretical and empirical research on the linkages. Recent studies confirming such linkages between the real economy and the financial sector include Lopez-Salido, Stein, and Zakrajsek (2017) for the United States and Amiti and Weinstein (2018) for Japan. For more details, including on the quantitative impact of such linkages in the FMM, see Appendix 1, "Quantification of the Negative Feedback-Loop Effect from the Financial Sector to the Real Economy," in Financial System Report - Annex, "The Financial Macro-econometric Model (FMM, March-2020 Version): Overview and Recent Developments" (published in August 2020).

²¹ Specifically, 17 banks (Mizuho Bank, MUFG Bank, Sumitomo Mitsui Banking Corporation, Mitsubishi UFJ Trust and Banking Corporation, Mizuho Trust and Banking Company, Sumitomo Mitsui Trust Bank, Gunma Bank, Chiba Bank, Bank of Yokohama, Hachijuni Bank, Hokkoku Bank, Shizuoka Bank, Shiga Bank, Chugoku Bank, Yamaguchi Bank, Iyo Bank, and Bank of Nagoya) are internationally active banks, while all others are domestic banks.

shinkin banks are subject to domestic capital adequacy requirements.

2.3. Estimation method

The FMM is estimated on an equation-by-equation basis, and many of the estimations are conducted using linear regression. This is a common approach for the estimation of large macro models.²² The data used in the estimation include official statistics such as SNA statistics, data provided by external vendors²³ such as granular data for the corporate sector, as well as the results of various surveys conducted by the BOJ and tabulations reported by individual banks. In principle, the estimates are based on data from as far back as possible in the past to as recent as possible, and the estimates are updated in time to the stress testing for the FSR. However, in cases where it is clear that the relationship between variables in the model deviates from the past due to temporary and special factors, separate measures are taken. For instance, for the estimation of the credit cost model, the end of the estimation period is set to fiscal 2019 because of the policy measures to support corporate financing in response to the Covid-19 pandemic from fiscal 2020.²⁴ Moreover, all data used in the FMM are in yen, and when taking the impact of exchange rates on items related to business in foreign currencies into account, the U.S. dollar/yen exchange rate sensitivity of items is estimated on a yen basis.

²² For instance, in the FRB's macroeconomic forecasting model, the FRB/US model, most of the model is estimated using equation-by-equation estimation, since it would be "infeasible" to estimate a very large number of equations simultaneously (Brayton, Laubach, and Reifschneider 2014). The same applies to the BOJ's macroeconomic forecasting model, the Q-JEM.

²³ For example, to calculate the ICR of each bank's borrower firms, financial information on individual firms and information on the banks with which they do business provided by Teikoku Databank is used.

²⁴ While the identification of structural changes requires a sufficiently large number of observations before and after such changes, in general, observations tend to be scarce especially after a structural change, so that in practice changes are identified by looking at the time-series behavior of the data and taking into account whether there have been financial or economic events that may have had a major impact on the data, meaning that to some extent the identification of structural changes is based on a subjective assessment. It is therefore necessary to systematically examine ex post whether such assessments were correct, using statistical methods. Relatedly, Japanese data suggest that there was a structural break in the relationship between certain variables around the year 2000, when there were a large number of mergers of financial institutions, and models employing those variables are estimated using data after the potential structural change.

3. Structure of the Model of the Domestic Banking Sector

As mentioned earlier, one of the most important variables in the FMM is the capital adequacy ratio of banks. In the domestic banking sector, the ratio is calculated by combining five different types of modules. The following explains the framework of the calculation and then provides details of each module.



Chart 3. Framework of the calculation of capital adequacy ratios

The investment and funding module (3.1 in Chart 3 above, described in detail in Section 3.1 below) is used to calculate developments in loans (in asset side) and funding (in liability side) in each scenario. For example, in a scenario in which domestic and foreign economic and financial conditions are firm, banks increase their lending and funding; conversely, in a scenario in which economic and financial conditions deteriorate, banks' lending -- and at the same time their funding -- decreases due to weak corporate demand in loan markets and tightening of banks' lending stance.

The periodic profit/loss module (3.2 in Chart 3, described in detail in Section 3.2) is used to calculate estimates for the items comprising net income, based on changes in investment and funding amounts, and the economic and financial environment assumed in the scenarios. Specifically, we estimate (i) pre-provision net revenue (PPNR) excluding trading income (=operating profits from core business), which refers to income from lending and securities investment-related businesses and fees and commissions, (ii) realized gains/losses on securities holdings, which refers to the amount of capital gains or losses realized from the securities investment business, and (iii) credit costs due to changes in the creditworthiness of borrower firms.²⁵ For example, in a scenario in which

²⁵ Credit costs are defined as Loan-loss provisions + Write-offs + Losses on nonperforming credit

the domestic and foreign economic and financial environment deteriorates, in many cases, the margins on loans and on investments in securities are squeezed, leading to the decline of PPNR, and the prices of risk assets fall, putting downward pressure on realized gains/losses on securities holdings through impairment losses and other factors. In addition, credit costs increase due to defaults and a deterioration in the creditworthiness of borrower firms. As a result of these factors, net income declines.

Valuation gains/losses on securities holdings, which are the capital gains/losses generated from the securities investment business that have not been realized, are calculated using the module for valuation gains/losses on securities holdings (3.3 in Chart 3, described in detail in Section 3.3 below). Specifically, valuation gains/losses on securities realized through sales, redemptions, and devaluation from the change in the market value of securities in each period.

Risk-weighted assets are calculated in the risk-weighted assets module (3.4 in Chart 3, details are provided in Section 3.4), which incorporates changes in risk-weighted assets due to changes in the creditworthiness of borrower firms, etc., as well as changes in the amount of lending and the market value of stockholdings.²⁶ Banks' capital adequacy ratios for the current period is calculated in the capital adequacy module (3.5 in Chart 3, see Section 3.5 for details) by adding the net income (after tax) obtained in the periodic profit/loss module and changes in valuation gains/losses on securities holdings to the amount of capital in the previous period, and then subtracting dividends. Finally, capital adequacy ratios are calculated by dividing the amount of capital by risk-weighted assets.²⁷

sales – Recoveries of write-offs.

²⁶ In this context, it should be noted that, in line with current regulations, the model for the risk-weighted assets of banks that have adopted the standardized approach is different from the model for those that have adopted the internal ratings approach.

²⁷ The capital adequacy ratio is calculated in accordance with Basel III (Basel Committee on Banking Supervision 2021) and the public notices issued by the Financial Services Agency in line with the Basel III regulations. Meanwhile, even when the simulation period includes a period in which the application of the finalized version of the Basel III regulations (Basel Committee on Banking Supervision 2017b) begins, the capital adequacy ratios are calculated based on the current regulations. Individual banks in the banking sector are modeled on a consolidated basis, including consolidated subsidiaries of the relevant bank. Because for consolidated subsidiaries the data necessary to identify risk factors is often limited, and it is difficult to model them in a manner that is linked to economic and financial conditions, asset and liability items are assumed to remain unchanged during the simulation period, and periodic profits/losses are assumed to be zero, so that they do not contribute to changes in the capital adequacy ratio in the simulation. However, in light of the losses incurred by subsidiaries in the bankruptcy of Sanyo Securities in 1997 and the bailout of American International Group (AIG) in 2008, modeling consolidated subsidiaries is an important future issue from the perspective of expanding the coverage of risks that financial institutions may face.

3.1. The investment and funding module

In the investment and funding module, changes in important investment and funding account items, such as items accounting for large amounts, are modeled separately, while the other balance sheet items are modeled together (Charts 4 and 5).²⁸ Moreover, domestic and foreign account items are modeled separately.²⁹ The reason is that it is assumed that the determinants of changes in domestic and foreign account items differ, even if the account items are the identical.

Starting with the asset side, asset items are classified into three categories: loans,³⁰ securities (on a book value basis), and other assets.

Investment	Funding		
A-1. Loans	B-1. Deposits		
A-2. Securities	B-2. Payables under repurchase agreements		
A-3. Trading securities	B-3. Negotiable certificates of deposit		
A-4. Money held in trust	B-4. Bonds		
A-5. Call loans	B-5. Call money		
A-6. Receivables under resale agreements	B-6. Payables under securities lending transactions		
A-7. Receivables under securities borrowing transactions	B-7. Bills sold		
A-8. Bills bought	B-8. Commercial papers		
A-9. Monetary claims bought	B-9. Borrowings		
A-10. Due from banks	B-9-1. Borrowings from BOJ		
A-11. Negotiable due from banks	B-9-2. Borrowings from other banks		
A-12. Foreign exchanges	B-10. Foreign exchanges		
A-13 The others	B-11. Bonds payable and bonds with share acquisition rights		
	B-12. The others		

Chart 5. Investment and funding module

Investment		Funding			
	Loans [A-1]		Domestic [(B-1)++(B-12)]		
	Domestic		Deposits [B-1]		
	Corporate loans (\star)	F	Borrowings from BOJ [B-9-1]		
	Retail loans (★)		Market fundings[(B-2)++(B-12) <excl. b-9-1="">]</excl.>		
	Loans for local governments		Foreign [(B-1)++(B-12)]		
	Foreign (\bigstar)		Deposits [B-1]		
	ecurities [A-2] he others [(A-3)++(A-13)]		Repo [(B-2)+(B-6)]		
			Market fundings [(B-3)++(B-12)]		

Note: The \star denotes items that are modeled as endogenous variables. The letters and numbers in square brackets show the correspondence to the investment and funding account items in Chart 4.

²⁸ Assets and liabilities other than those in the investment and funding accounts, such as fixed assets, are assumed to remain unchanged from the end of the pre-simulation period.

²⁹ "Domestic" refers to the accounts of the domestic business sector, while "foreign" refers to the accounts of the international business sector (the same applies below).

 $^{^{30}}$ When we refer to loans in this paper, we are referring to the amount of loans outstanding (the same applies below).

Of these, only loans are assumed to fluctuate in response to changes in macroeconomic variables in the scenarios, while securities and other assets are assumed to remain unchanged in book value terms from the actual values at the end of the pre-simulation period throughout the simulation period, regardless of the scenario. Therefore, the value of total assets during the simulation period is the value at the end of the pre-simulation period plus the change in loans during the simulation period.³¹ In addition, it is assumed that the composition of borrowers of loans remains unchanged from the actual values at the end of the pre-simulation period. Proceeds from securities are also assumed to be reinvested in the same type of securities when they reach maturity, so that both the total amount and the composition of securities holdings are assumed to remain unchanged during the simulation period.³²

Loans are composed of domestic loans to corporations, individuals, and local governments, and foreign loans. With the exception of loans to local governments, which are assumed to remain unchanged from the pre-simulation period values in all scenarios, each of these are specified such that they are influenced by the economic and financial conditions to differing degrees. Moreover, the FMM is specified such that loans decline in line with the deterioration in the real economy,³³ and that this decline in loans spills over to the domestic economy, with the decline in banks' corporate loans negatively affecting firms' investment in the economy and the decline in loans to individuals negatively affecting private consumption.

On the liabilities side, funding is divided into domestic and foreign funding and is specified such that domestic funding fluctuates in line with changes in domestic loans and foreign funding fluctuates in lines with changes in foreign loans.³⁴ In this case, it is

³¹ From the perspective of taking the detailed linkages between the real economy and the banking sector into account, it would be desirable to model securities holdings in a manner such that their value depends on financial conditions, etc. This is an issue to be addressed in the future.

³² The assumption that the total amount and composition of financial institutions' balance sheet items remain unchanged during the simulation period is called the "static balance sheet assumption" and is widely used in macro stress testing as a regulatory tool by supervisors. Key reasons are that the assumption facilitates interpretation and comparison of results and eliminates the possibility of an upward bias in capital adequacy ratio estimates due to arbitrary and subjective assumptions on balance sheet dynamics (Basel Committee on Banking Supervision 2017a, Board of Governors of the Federal Reserve System 2021, European Banking Authority 2021).

³³ In the early stages of major recessions, such as during the global financial crisis of 2008 and the outbreak of the Covid-19 pandemic in 2020, lending by financial institutions relative to GDP temporarily increased in some cases due to firms' use of credit lines (Berrospide, Meisenzahl, and Sullivan 2012, Acharya, Engle, and Steffen 2021, and Aoki et al. 2021). However, since these loans were subsequently repaid relatively quickly, we use time dummies in our loan model estimates to control for fluctuations in lending in the immediate post-crisis period.

³⁴ The specification that the amount of liabilities is passively determined after the amount of assets is

assumed that the composition in terms of deposits, loans from the BOJ, and market-based funding among the domestic funding and the composition in terms of deposits, repos, and market-based funding among the foreign funding remain unchanged from the current compositions. The following subsections describe the specifications of the functions for domestic corporate loans, domestic loans to individuals, and foreign loans.

Model for domestic corporate loans

Domestic corporate loans are specified such that they are explained by macroeconomic and bank-specific factors that are thought to affect lending.³⁵ Specifically, the output gap,³⁶ the expected growth rate (over the next three years),³⁷ the population growth rate,³⁸ and the rate of change of land prices³⁹ are used as variables to capture changes in macroeconomic demand for loans by corporations. This specification is to capture changes in the volume of lending due to changes in firms' cash flows on hand, expected returns on investment, and the value of collateral held by firms. The lower these explanatory variables are, the lower is firms' demand in the loan market and the lower is the amount of loans.

The variables used to capture the lending stance of banks - the suppliers of loans - are

determined first implies that banks can always raise, both domestically and abroad, funds equal to the amount of assets; in other words, it does not assume extremely severe liquidity stress that would force banks to make large changes in the amount of assets in a short period of time as the amount of liabilities changes. This is due to the fact that the FMM focuses on examining the adequacy of banks' capital and not on liquidity risks, while the importance of liquidity risk was highlighted by Basel Committee on Banking Supervision (2013). The models used by the FRB and the ECB for stress testing (see Appendix 6) also exclude liquidity risk, while the IMF uses separate models for capital adequacy and liquidity stress testing (Adrian, Morsink, and Schumacher 2020). The FMM does not completely abstract from liquidity-related stresses but rather incorporates a mechanism whereby stresses in banks' financing can reduce capital adequacy ratios by increasing funding costs and squeezing periodic profits.

³⁵ Since standard economics assume that loans and interest rates are determined endogenously and simultaneously, it cannot be ruled out that in this specification the parameter estimates in the equation may be subject to simultaneous equation bias. For this reason, the simulation results based on this model should be interpreted with a degree of caution.

³⁶ The historical values of the output gap are estimated by the BOJ, and the forecast values for the simulation period are prepared using potential GDP estimated by applying a Hodrick-Prescott (HP) filter to the GDP scenario. Level adjustments are made when connecting the forecast values to the historical values.

³⁷ Actual figures for the pre-simulation period are based on the average for all industries of responses to the Cabinet Office's "Annual Survey of Corporate Behavior."

³⁸ The National Institute of Population and Social Security Research's estimates and projections of the working-age population (those aged 15 years and older) are used.

³⁹ For land prices, the Japan Real Estate Institute's urban land price index is used. The simulation period is specified in such a way that the ratio of land prices to nominal GDP converges to the historical average.

their capital adequacy ratios and net income return on assets (ROA).⁴⁰ The specification is such that the lower the capital adequacy ratio and the lower the net income ROA, the more loans decrease.⁴¹ Moreover, the amount of loans falls in a nonlinear fashion⁴² once the capital adequacy ratio falls below a certain threshold⁴³ and net income ROA turns red.

Another explanatory variable for loans used in the model is lending interest rates. A priori, the impact of lending interest rates on loans is ambiguous. For example, on the demand side, an increase in lending interest rates is expected to reduce firms' demand for borrowing, while on the supply side, banks are expected to take a more aggressive lending stance due to the higher lending interest rates. In practice, the estimation results suggest that an increase in lending interest rates reduces lending on a net basis.

Specification of the model for domestic corporate loans

Domestic corporate loans_i [y/y chg.] = $\alpha_1 \times \text{Output gap} + \alpha_2 \times \text{Expected economic growth rate}$ + $\alpha_3 \times \text{Population growth rate} + \alpha_4 \times \text{Growth rate of land prices}$ + $\alpha_5 \times (\text{Capital adequacy ratio}_i - \text{Threshold}_i) \times (1 + \gamma_1 \times \text{Dummy}_{\text{CAR}_i < \text{threshold}_i})$ + $\alpha_6 \times ROA_i \times (1 + \gamma_2 \times \text{Dummy}_{\text{Net income ROA}_i < 0})$ + $\alpha_7 \times \text{Domestic lending interest rate}_i$ [1-quarter lag, chg. from previous year] +Fixed effect_i + Constant -- *i* denotes individual banks, hereafter.

Model for domestic loans to individuals

While the explanatory variables in the model for domestic loans to individuals are the same as those in the model for domestic loans to corporations, they are estimated in separate models since the sensitivities are not necessarily the same. In fact, the estimated coefficients do differ; for example, the sensitivity of loans to individuals to the output gap

⁴⁰ Ratio of net income to total assets = (After-tax) net income/Total assets.

⁴¹ This follows preceding empirical studies showing that lower capital adequacy ratios reduce loans. See Peek and Rosengren (1997, 2000) and Amiti and Weinstein (2018) for studies on Japanese banks and Berrospide and Edge (2010) for a study on U.S. banks.

⁴² Such nonlinear relationships for both the net income ROA and capital adequacy ratios are found in panel estimations for banks presented in the October 2015 FSR and the October 2016 Annex, "Macro Stress Testing in the *Financial System Report* (October 2016)" (released in October 2016).

 $^{^{43}}$ The threshold for internationally active banks is a total capital adequacy ratio of 8 percent, while the threshold for domestic banks and *shinkin* banks is a capital ratio of 4 percent (the same applies to the function for domestic loans to individuals and the function for international loans).

is smaller than that of loans to corporations.

Specification of the model for domestic loans to individuals

Domestic household loans_i [y/y chg.] = $\alpha_1 \times \text{Output gap} + \alpha_2 \times \text{Expected economic growth rate}$ + $\alpha_3 \times \text{Population growth rate} + \alpha_4 \times \text{Growth rate of land prices}$ + $\alpha_5 \times (\text{Capital adequacy ratio}_i - \text{Threshold}_i) \times (1 + \gamma_1 \times \text{Dummy}_{\text{CAR}_i < \text{threshold}_i})$ + $\alpha_6 \times ROA_i \times (1 + \gamma_2 \times \text{Dummy}_{\text{Net income ROA}_i < 0})$ + $\alpha_7 \times \text{Domestic lending interest rate}_i$ [1-quarter lag, chg. from previous year] +Fixed effect_i + Constant

Model for foreign loans

The model for foreign loans uses as explanatory variables the growth trend in loans specific to each bank and macroeconomic factors that are regarded to be determinants of fluctuations in loans. In this regard, the output gap for foreign economies⁴⁴ is used as a variable to capture changes in the demand for loans by foreign firms, and the more negative the gap, the lower the demand in the loan market and the less loans are extended. Moreover, as with loans to domestic borrowers, the capital adequacy ratios of banks is used as a variable to capture their lending stance, and the dollar/yen exchange rate is included as an explanatory variable to account for changes in the yen value of loans due to exchange rate fluctuations.

Specification of the model for foreign loans

Foreign loans_i [y/y chg.] = $\alpha_1 \times$ Foreign output gap [chg. from previous year] + $\alpha_2 \times$ (Capital adequacy ratio_i – Threshold_i) + $\alpha_3 \times$ Exchange rates [USD/JPY, y/y chg.] + Fixed effect_i + Constant

3.2. The periodic profit/loss module

For periodic profits/losses, of the various items in banks' income statements (Chart 6), those that make a substantial contribution to overall changes in periodic profits/losses, such as items related to their lending and securities investment business, are modeled. In this regard, commission business-related items are modeled only for major banks, and the other items are assumed to remain constant in all scenarios, in line with historical averages,

⁴⁴ Actual values for the pre-simulation period are based on the IMF's World Economic Outlook. Forecasts for the simulation period are prepared using potential GDP estimated by applying HP filters to the GDP scenarios for foreign economies.

etc.⁴⁵ For the FMM, these items related to net periodic profits/losses are classified into three top-level account items (Chart 7).

Revenues	Expenses		
A. Ordinary income (A-1)++(A-5)	B. Ordinary expenses (B-1) ++(B-6)		
A-1. Interest income	B-1. Interest expenses		
A-1-1. Interest on loans	B-1-1. Domestic business		
A-1-2. Interest and dividends on securities	D 1 2 Internetic allowing as		
A-1-3. The others	B-1-2. International business		
A-2. Fees and commissions	B-2. Fees and commissions payments		
A-3. Trading income	B-3. Trading expenses		
A-4. Other ordinary income	B-4. Other ordinary expenses		
A-4-1. Gain on sales or redemption of bonds	B-4-1. Loss on sales, redemption or devaluation of bonds		
	B-4-2. The others		
A-4-2. The others	B-5. General and administrative expenses		
A-5. Other income	B-6. Other expenses		
A-5-1. Gain on sales of stocks	B-6-1. Loss on sales or devaluation of stocks		
A-5-2. Reversal of allowance for loan losses/	B-6-2. Provision of allowance for loan losses/		
Recoveries of written off claims	Written-off of loans		
A-5-3. The others	B-6-3. The others		
C. Ordinary profit (loss) (A)-(B)			
D. Extraordinary income	E. Extraordinary losses		
F. Profit (loss) before tax (C)+(D)-(E)			
	G. Income taxes		
H. Profit (loss) (E)-(G)			

Chart 6. Income statement items

The first of these is PPNR excluding trading income, which represents banks' fundamental profitability and changes over time depending on factors such as lending interest rates, yields on securities holdings, funding rates, and the size of loan portfolios. The second is realized gains/losses on securities holdings, which consist of realized gains/losses on bonds, stocks, and other securities and fluctuate with changes in the market value of securities holdings. The third is credit costs, which vary depending on financial conditions in the corporate sector. The sum of PPNR excluding trading income and gains/losses on securities holdings minus credit costs (a cost item) equals net income before taxes, and subtracting tax expenses from this net income yields the after-tax income.⁴⁶ The following subsections explain the structure of the models used for the different items, i.e., PPNR excluding trading income, realized gains/losses on securities holdings.

⁴⁵ The use of historical averages, etc., instead of actual values just before the simulation period is due to the large variation in some items over time.

 $^{^{46}}$ Tax expenses are calculated assuming an effective tax rate of 40 percent for banks and 30 percent for *shinkin* banks provided that their net income before taxes is in the black. On the other hand, the effective tax rate is assumed to be zero for banks whose net income is in the red.





3.2.1. PPNR excluding trading income

PPNR excluding trading income is defined as the sum of net interest income⁴⁷ and net non-interest income⁴⁸ minus general and administrative expenses.⁴⁹ As shown in Chart 8, PPNR excluding trading income is calculated using the results of several models.

For net interest income-related items, interest rates are modeled both for income and expense items, and the amount of net interest income is calculated by combining the simulated values of the interest rates with those of the amount of loans and other items obtained from the investment and funding module discussed earlier. For non-interest income, which mainly derives from fee business, the amount of income itself is modeled.

⁴⁷ Net interest income = Interest income – Interest expenses.

⁴⁸ Net non-interest income = Net fees and commissions + Net trading income + Other net non-interest income – Realized gains/losses on bondholdings. Net fees and commissions are fees and commissions received as compensation for services provided to customers in the lending, securities, and foreign exchange business, minus fees and commissions paid. Net trading income are the difference between income and expenses arising from securities transactions and derivatives transactions in the trading account (used for trading purposes). Other net ordinary income are profits from banks' basic business operations other than net interest income, net fees and commissions, and net trading income (e.g., gains/losses from the sale and purchase of foreign exchange).

⁴⁹ General and administrative expenses are incurred in conducting business activities, including personnel expenses such as salaries and remuneration of executives and employees, property expenses such as depreciation of tangible and intangible fixed assets and advertising expenses, and taxes such as consumption tax.

Chart 8. PPNR excluding trading income

PPNR (excl. trading income)			
Net interest income $[(A-1-1)+(A-1-2)+(A-1-3)-(B-1)]$			
Loan interest income [A-1-1]			
Domestic : Domestic lending interest rate (\star)×Domestic loans			
Foreign: Foreign lending interest rate (*)×Foreign loans			
Interest and dividends on securities [A-1-2] : Yields on securities × Securities holdings			
The others [A-1-3] : Yields on other assets ×Other assets holdings			
Interest expenses [B-1]			
Domestic: Domestic funding rate (★)×Domestic fundings [B-1-1]			
Foreign: Foreign funding rate (★)×Foreign fundings [B-1-2]			
Net non-interest income[$(A-2)-(B-2)+(A-3)-(B-3)+(A-4)-(B-4)-((A-4-1)-(B-4-1))$]			
Net fees and commissions (\bigstar) [(A-2)–(B-2)]			
The others (\bigstar) (Net trading income + Other net ordinary income -realized gains/losses on bondholdings)			
[(A-3)-(B-3)+(A-4)-(B-4)-((A-4-1)-(B-4-1))]			
General and administrative expenses [B-5]			
Domestic			
Foreign			

Note: The \star denotes items that are modeled as endogenous variables. The letters and numbers in square brackets show the correspondence to the income statement items in Chart 6.

Starting with net interest income, this is calculated as the sum of loan-related interest income, interest and dividends on securities, and other income⁵⁰ (i.e., income items) minus funding costs (i.e., expense items). Loan-related interest income is calculated as the sum of domestic loan-related interest income, which is calculated by multiplying domestic loans by domestic lending interest rates, and foreign loan-related interest income, which is calculated by multiplying foreign loans by the foreign lending interest rates. Interest and dividends on securities are calculated as securities holdings multiplied by the yield on securities. Securities yields include bond yields and dividend yields on stocks, etc. Bond yields are calculated based on each remaining maturity using data⁵¹ on each bank's maturity ladder (amount of bondholdings by remaining period) at the end of the pre-simulation period.⁵²

⁵⁰ Other income includes income from call loans and interest received on due from other banks.

⁵¹ Regarding bond maturity ladder of Japanese banks, the BOJ conducts a quarterly survey of banks that have a current account at the BOJ. From this survey, the BOJ obtains information on banks' bondholdings by maturity and yields on those bondholdings.

⁵² For bond interest income, the average effective yield is used as an approximation of the yield at the time of acquisition. In addition, to reflect the maturity composition of the outstanding amount of bonds in the market and heterogeneity in the maturity composition of the outstanding amount held by banks, (1) banks' composition of bondholdings by remaining maturity is multiplied by (2) the share of bondholdings by original maturity to calculate the composition of outstanding bonds by remaining and original maturity. (2) is assumed to be equal to the share calculated using data on outstanding bond issuances in the market (excluding the holdings by the BOJ). The average effective yield is then calculated using the composition of outstanding bonds. The yield on the proceeds from bonds that

Dividend yield from stocks and other assets is assumed to remain unchanged from the actual values just before the simulation period.⁵³ Other investment income, calculated by multiplying other asset holdings by the yields on other assets, is assumed to be linked to changes in funding costs so that the profit margin on other assets remains constant over the simulation period. Funding costs are the sum of domestic and foreign funding costs, calculated as the amount of domestic and foreign funding multiplied by the corresponding funding interest rates.

For net non-interest income, net fees and commissions and other net non-interest income are modeled separately. Other net non-interest income is defined as the sum of net trading income and other net ordinary income (excluding realized gains/losses on securities holdings).

General and administrative expenses are calculated as the sum of domestic and foreign expenses. In all scenarios, domestic expenses are assumed to remain unchanged from the actual values just before the simulation period. Conversely, foreign expenses, which have been on an upward trend, are assumed to fluctuate in line with the outstanding amount of foreign funding in the baseline scenario, because they have been stable as a percentage of the outstanding amount of foreign funding. In the stress scenarios, foreign-currency expenses in local currency terms are assumed to be constant, and a downturn in foreign economies during the simulation period itself is assumed to have no effect on the foreign expenses. This is based on the assumption that it is difficult for banks to reduce expenses (in local currency) in the short run even if their foreign business shrinks due to recession.

The following subsections describe the domestic lending interest rate model, the foreign lending interest rate model, the domestic funding interest rate model, the foreign funding interest rate model, the model for net fees and commissions, and the model for other net non-interest income.

Domestic lending interest rate model

Domestic lending interest rates are specified as changing in tandem with changes in market interest rates, changes in credit premiums associated with changes in borrower firms' creditworthiness, and changes in margins associated with changes in the supply-

have matured and are reinvested is specified such that it is linked to interest rates on government bonds with the corresponding remaining and original maturity assumed in each scenario. In addition, it is assumed that during the simulation period banks reinvest the proceeds in line with the investment shares at the end of pre-simulation period, so that banks' current investment policy is reflected in the term composition of their bondholdings going forward.

⁵³ As a result, the amount of dividends in the simulation fluctuates to the same extent as stock prices.

demand conditions. First, the linkage with changes in market interest rates is expressed by using the (short-term) domestic funding rate and the difference between short- and long-term interest rates (5-year minus 3-month rates) as explanatory variables.⁵⁴ Next, the nonperforming loan (NPL) ratio is included as an explanatory variable to proxy credit premiums associated with changes in borrower firms' creditworthiness, and it is assumed that banks add a premium for credit risk to domestic lending interest rates when the NPL ratio increases. Finally, the loan demand index,⁵⁵ which is defined as the ratio of the number of borrowing firms to the number of banks' branches in a prefecture, is used as an explanatory variable to proxy for changes in the supply-demand conditions in the loan market.⁵⁶ In the domestic lending interest rate model, the sensitivity of lending interest rates to the different variables is assumed to be identical across banks.

⁵⁴ The estimates of the degree of linkage between these variables and domestic lending interest rates are all below one. This is consistent with the results obtained by Kitamura, Muto, and Takei (2016), who examine the spillover effect from market interest rates to lending rates at Japanese banks.

⁵⁵ The loan demand index is calculated as the number of borrowing firms divided by the number of banks' branches in each prefecture and can be interpreted as an indicator of the degree of market concentration on the supply side (banks) in the local loan market. The purpose of including this variable in the function of domestic interest rate is to capture the mechanism by which long-term structural factors, such as the shrinking of the population and decline in the number of borrowing firms, exert downward pressure on lending interest rates through a continuous easing of supply-demand conditions for loans. However, it should be noted that (1) with the increase in so-called crossprefectural lending by regional banks (Ozaki et al. 2019), i.e., lending by banks in prefectures outside that in which their head office is located, potential competitors in a particular prefecture's loan market may include not only banks located in that prefecture but also those in other prefectures; (2) with the advances in digitalization, the link between the number of physical branches and the degree of concentration in the loan market may weaken; and (3) the sign of the correlation between the reduction in the number of branches and lending interest rates is a priori ambiguous. In particular with regard to the third point, if the reduction in the number of branches reduces the degree of competition among banks, this could lead to higher lending interest rates, but if banks reduce the number of branches (to cut fixed costs) in order to create room for lowering lending interest rates and increasing their competitiveness, this could lead to lower lending interest rates. Estimation results suggest that the former effect outweighs the latter, since the loan demand index is positively correlated with lending interest rates. See Chapter VI, Section B, of the October 2019 FSR for a discussion of the effect of different assumptions about the dynamics of the loan demand index during the simulation period on the predictions about profitability and capital adequacy ratios of banks. When calculating the loan demand index, we use the prefecture where the head office is located in the case of regional and shinkin banks, while for large banks with a nationwide presence, we use national values (average of all prefectures). Finally, it should be noted that apart from those used in the FMM, there are several other indicators of supply-side market concentration in loan markets (see van Leuvensteijn et al. 2011 for details).

⁵⁶ The positive correlation between the degree of market concentration among banks in the loan market and loan spreads has been shown theoretically in the Klein-Monti model (Klein 1971 and Monti 1972), the standard lending interest rate model, and has also been found in empirical studies. See, for example, van Leuvensteijn et al. (2013), for an analysis using eurozone data.

Specification of the domestic lending interest rate model
Domestic lending interest rate _i
$= \alpha_1 \times \text{Domestic funding rate}_i + \alpha_2 \times \text{Term spread [5-year - 3-month]}$
$+\alpha_3 \times \text{Non-performing loan ratio}_i + \alpha_4 \times \text{Loan demand index}_i$
+Fixed effect _i + Constant
Loan demand index _i
= The number of borrowing firms in the prefecture where i 's head office is located
The number of branches in the prefecture where <i>i</i> 's head office is located

Foreign lending interest rate model

The model for foreign lending interest rates is similar to that for domestic lending interest rates but somewhat simplified. While the explanatory variables for foreign lending interest rates include funding rates as in the model for domestic loans interest rates, the foreign output gap instead of the loan demand index is used to capture supply and demand conditions in the loan markets.⁵⁷

Specification of the foreign lending interest rate model

Foreign lending interest rate _i = $\alpha_1 \times$ Foreign funding rate _i
$+\alpha_2 \times$ Foreign output gap + Fixed effect _i + Constant

Model for domestic funding rates

Domestic funding rates are the weighted average of the interest rate levels projected from the model for the three sources of funding, namely, (1) deposits, (2) borrowings from the BOJ,⁵⁸ and (3) market-based funding (which is the sum of all funding sources other than (1) and (2)), using the shares of these funding sources at the end of the pre-simulation period as weights. In other words, it is assumed that in the event of stress there is no major shift in funding sources in terms of (1) through (3). While in the functions for both (1) and (2) the only explanatory variable is the three-month JGB yield, estimating the two functions makes it possible to capture differences in the degree of spillover to the interest rates for the two funding sources. The interest rate for (3) is specified such that it is affected by the level of banks' capital adequacy ratio in addition to the three-month JGB

⁵⁷ On the one hand, a deterioration in the foreign output gap may widen credit premiums due to a deterioration in the creditworthiness of borrower firms. On the other hand, it may decrease demand for funds by firms in the loan market, which puts downward pressure on the lending interest rate. Since the estimated coefficient for the foreign output gap is positive, this suggests that the latter effect is larger than the former.

⁵⁸ Borrowing from the BOJ refers to the borrowing of funds from the BOJ by banks such as through various operations including the Complementary Lending Facility and the Loan Support Program.

yield.

(1)Deposit interest rate_i = $\alpha_1 \times$ Short-term interest rate [3-month T-Bills] +Fixed effect_{*i*} + Constant (2)Borrowing interest rate from $BOJ_i =$ $\alpha_1 \times \text{Short-term interest rate [3-month T-Bills]} + Fixed effect_i + Constant$ (3)Market-based funding rate_{*i*} = $\alpha_1 \times$ Short-term interest rate [3-month T-Bills] $+\alpha_2 \times \text{Capital adequacy ratio}_i + \text{Fixed effect}_i + \text{Constant}$

Specification of the model for domestic funding rates

Model for foreign funding rates

Foreign funding rates are the weighted average of the interest rate levels projected from the model for the three sources of funding, i.e., (1) deposits, (2) repo funding, and (3) the other market-based funding except repo funding, including certificates of deposit/commercial paper (CD/CP) and FX and foreign currency swaps (other marketbased funding again is the sum of all funding sources other than (1) and (2)), using the shares of these funding sources at the end of the pre-simulation period as weights. Of these, deposit interest rates are specified as being linked to three-month U.S. Treasury bill yields and individual banks' capital adequacy ratio, 59 and a decline in banks' creditworthiness leads to an increase in the funding rate they face. Next, the repo rates are also specified as being linked to three-month U.S. Treasury bill yields, but since repo transactions are secured transactions, variables representing banks' creditworthiness are not included in the explanatory variables. Finally, (3) the other market-based funding rates are specified such that, in addition to banks' capital adequacy ratio, they are determined by the sum of the (three-month) dollar LIBOR and the (three-month) dollar funding premium,⁶⁰ and are affected by stresses in foreign currency funding conditions as well as banks' creditworthiness.⁶¹ In the stress scenarios, the increase in the other market-based

⁵⁹ In order to ensure that sufficiently long time series are available, the total capital adequacy ratio (for internationally active banks) and the capital adequacy ratio (for domestic banks) are used, like in the loan models.

⁶⁰ The dollar funding premium in the FMM is defined as the difference between the cost of dollar funding using foreign exchange swaps and the (three-month) dollar LIBOR.

⁶¹ The specification of foreign currency funding rates is based on the observation that the stress in repo funding in U.S. and European financial markets was relatively modest during past financial crises (see, e.g., Krishnamurthy, Nagel, and Orlov 2014, Copeland, Martin, and Walker 2014, and Mancini, Ranaldo, and Wrampelmeyer 2016), while market-based funding other than repo transactions, such as CDs/CP and FX and currency swaps, was under considerable stress due to the emergence of counterparty risk and other factors (see, e.g., Baba and Packer 2009, Afonso, Kovner, and Schoar 2011, Covitz, Liang, and Suarez 2013, and Chernenko and Sunderam 2014).

funding rates tends to be particularly large, and the higher a bank's share of the other market-based funding, the more likely is its overall foreign funding rate to rise.

Specification of the model for foreign funding rates
(1)Deposit interest rate _i = $\alpha_1 \times$ Short-term interest rate [3-month T-Bills]
$+\alpha_2 \times \text{Capital}$ adequacy ratio _i + Fixed effect _i + Constant
(2)Repo rate _i = $\alpha_1 \times$ Short-term interest rate [3-month T-Bills]
+Fixed $effect_i$ + Constant
(3)The other market-based funding $rate_i =$
$\alpha_1 \times (U.S.dollar LIBOR + U.S.dollar funding premium)$
$+\alpha_2 \times \text{Capital}$ adequacy ratio _i + Fixed effect _i + Constant

Model for net fees and commissions

Net fees and commissions are modeled for major banks only, and the model is specified such that fees and commissions income from services such as foreign exchange business and trust and insurance sales⁶² are explained by variables representing economic and financial conditions, such as the output gap, stock prices, and exchange rates.⁶³ In addition, given the strong increase in this account from the late 1990s to the 2000s due to deregulation,⁶⁴ a time trend⁶⁵ for this period is added as an explanatory variable to represent deregulation.

Specification of the model for net fees and commissions

Net fees and commissions to total assets (Major banks, excl. trust fee)*i* $= \alpha_1 \times \text{Output gap} + \alpha_2 \times \text{Stock price [y/y chg.]}$ $+\alpha_3 \times \text{Exchange}$ rates to total assets [USD/JPY] $+\alpha_4 \times \text{Deregulation}$ trend +Fixed effect_{*i*} + Constant

Model for other net non-interest income

Other net non-interest income is also modeled for major banks only. The types of business

⁶² Trust fees, which are fees received as compensation for the management and administration of investment trusts, are also included in net fees and commissions.

⁶³ For major banks, the exchange rate is included as an explanatory variable only for internationally active banks to take into account the impact of exchange rate changes on the yen value of foreign net fees and commissions.

⁶⁴ Specifically, banks have been able to own securities subsidiaries since 1993 and handle over-thecounter sales of investment trusts since 1998, and the ban on insurance sales was completely lifted in 2007.

⁶⁵ Specifically, a quarterly time trend is included as an explanatory variable for the period 1997–2007.

included in this account are diverse; however, reflecting their size and variability and their high correlation with macroeconomic variables that represent economic and financial conditions, only (1) gains/losses on interest rate swap transactions and (2) gains/losses on currency and foreign exchange swap transactions (consisting of the sum of gains/losses on currency swap transactions and gains/losses on foreign exchange trading) are assumed to vary with economic and financial conditions. Meanwhile, the other items, including gains/losses on financial derivatives, in all scenarios are assumed to remain unchanged from their historical averages.

For the modeling, variables that show a high correlation with (1) and (2) in the past are selected as common explanatory variables for all banks; however, given the extremely high heterogeneity among banks in the types of business operations included in this account, the estimation was conducted using a specification that allows the coefficients to differ across banks. Specifically, for (1) gains/losses on interest rate swap transactions, the quarter-on-quarter difference in the (three-month) dollar LIBOR, which is negatively correlated with gains/losses on interest rate swap transactions, is used as an explanatory variable, ⁶⁶ while for (2) gains/losses on currency swap and foreign exchange swap transactions, the U.S.-Japan interest rate differential (three-month U.S. interest rates minus three-month Japanese interest rates), which is positively correlated with gains/losses on currency swap and foreign exchange swap transactions, etc., and the output gap are used as explanatory variables.⁶⁷

⁶⁶ The (three-month) dollar LIBOR is used as a proxy for developments in global interest rates, including yen interest rates. In practice, looking at the yen interest rate swap positions of major banks, that of fixed interest receipts/floating interest payments consistently exceed that of fixed interest payments/floating interest receipts, which may be one of the reasons for the increase in profits during periods of declining interest rates.

⁶⁷ The positive correlation with the US-Japan interest rate differential suggests that Japanese banks' positions in dollar interest receipts/yen interest payments are larger than their positions in dollar interest rate payments/yen interest receipts. The output gap is used as a variable to capture the volume of transactions in the customer market.

Specification of the model for other net non-interest income

$(1) \frac{\text{Gains/losses on interest rate swap}_i}{\text{Funding amount}_i}$
$= \alpha_i \times U.S.$ dollar LIBOR [chg. from previous quarter] + Fixed effect _i + Constant
$(2) \frac{\text{Gains/losses on currency swap}_i + \text{Foreign exchange transaction}_i}{\text{Funding amount}_i}$
$= \alpha_{1,i} \times U.$ SJapan 3-month interest rate differential [chg.from previous quarter]
$+\alpha_2 \times \text{Output gap} + \text{Fixed effect}_i + \text{Constant}$

3.2.2. Realized gains/losses on securities

Realized gains/losses on securities holdings are the sum of realized gains/losses on bondholdings and realized gains/losses on stockholdings (Chart 9). ⁶⁸ Realized gains/losses on bondholdings correspond to the balance of five accounts for bonds⁶⁹ and realized gains/losses on stockholdings correspond to the balance of three accounts for stocks and other securities.⁷⁰

⁶⁸ To simplify the calculation, if the financial variable used as a risk factor is a price, the item is included in realized gains/losses on stockholdings, and if it is an interest rate (or spread), the item is included in realized gains/losses on bondholdings.

⁶⁹ Specifically, realized gains/losses on bondholdings are calculated as follows: Gains on sales of bonds + Gains on redemption of bonds – Losses on sales of bonds – Losses on redemption of bonds – Losses on devaluation of bonds. Gains/losses on the sale of bonds are the difference between the amount realized on the sale of bonds and their book value. While gains/losses on the redemption of bonds to which the amortized cost method is not applied are the difference between the redemption amount at maturity and the book value, gains/losses on the redemption of bonds in the FMM are zero, since it is assumed that the amortized cost method is applied during the simulation period. Losses on the devaluation of bonds consist of (1) the valuation losses on available-for-sale securities for banks that have adopted an accounting method under which only valuation gains are included directly in net assets and (2) impairment losses on held-to-maturity bonds and available-for-sale securities. Since in the FMM it is assumed that all banks adopt an accounting method under which both valuation gains and losses are included directly in net assets, (1) is zero, while (2) varies depending on the magnitude of the stress.

⁷⁰ Specifically, realized gains/losses on stockholdings are calculated as follows: Gains on sales of stocks – Losses on sales of stocks – Losses on devaluation of stocks. Gains/losses on sales of stocks are gains/losses on the sales of securities other than bonds. Losses on the devaluation of stocks correspond to (1) the valuation losses on available-for-sale securities for banks that adopt an accounting method under which only valuation gains are included directly in net assets and (2) impairment losses on stocks and other securities. Since in the FMM it is assumed that all banks adopt an accounting method under which both valuation gains and losses are included directly in net assets during the simulation period, (1) is zero, while (2) varies depending on the magnitude of the stress.

Chart 9. Securities in the module for realized gains/losses on securities holdings Realized gains/losses on securities holdings [((A-5-1)-(B-6-1))+((A-4-1)-(B-4-1))]

ealize	ed gains/losses on bondholdings [(A-4-1)-(B-4-1)]
G	overnment bonds, etc
D	omestic credit products
	Corporate bonds (AAA, AA, A and BBB rated or lower, incl. bank debentures)
F	oreign bonds
	Dollar-denominated
	Euro-denominated
F	oreign credit products
	Coporate bonds (AAA, AA, A, BBB, BB, B and CCC rated or lower, incl. bank debentures)
	CLO (AAA, AA, A, BBB and BB rated or lower)
	CMBS (AAA, AA, A and BBB rated or lower)
	RMBS
	ABS (AAA and AA rated or lower)
	Direct lending funds
	Bank loan funds (BB rated or higher and B rated or lower)
ealize	a gains/losses on stockholdings [(A-5-1)-(B-6-1)]
D	omestic stocks and stock investment trusts
D	omestic investment funds and alternative investments
	Bond investment funds
	Credit investment funds
	Real estate investment funds
	Balanced funds
	Hedge funds
F	oreign stocks and stock investment trusts
F	oreign investment funds and alternative investments
	Real estate investment funds
	Hedge funds
	Private equity

Note: The letters and numbers in square brackets show the correspondence to the income statement items in Chart 6.

Realized gains/losses on securities holdings consist of (1) gains/losses on sales of securities (gains/losses on sales of bondholdings and gains/losses on sales of stockholdings) and (2) losses when the market value of securities falls substantially below their book value and the losses are written off (devaluation of bondholdings and stockholdings). The value of each term is a function of the valuation gains/losses on securities holdings at the end of the previous period (i.e., at the beginning of the current period), and the values calculated in the module for valuation gains/losses on securities holdings are used as arguments in the calculation. Meanwhile, securities sales and

devaluation during a period are deducted from the valuation gains/losses on securities at the beginning of the period to arrive at the valuation gains/losses at the end of the current period (i.e., at the beginning of the next period).

The following paragraphs and Chart 10 explain the items for which we calculate the realized gains/losses and the calculation method. The calculation of realized gains/losses is based on the valuation gains/losses for each type of securities, estimated by the model for valuation gains/losses on securities holdings.

First, as for (2) the devaluation of bondholdings and stockholdings, we calculate the devaluation of credit investments (included in bondholdings) and alternative investments (included in stockholdings) during the simulation period for each type of security. In this calculation, when the market-to-book ratio declines and valuation losses expand to 50 percent or more of the book value, it is assumed that banks sell their holdings to cut losses.

		(1)Sales	(2)Devaluation	(3)Redemption
Stocks	Stocks and stock investment trusts	Coloulated for stools alto esther	Calculated for stocks and stock investment trusts, altogether	
	Investment funds and alternative investments	Calculated for stocks allogether	Calculated by item	
Bonds	Government bonds	Calculated for bonds altogether	Assumed to be zero	Assumed to be zero (Applying amotized cost method)
	Credit investments (corporate bonds, securitized products)	Not covered	Calculated by item and rating just before simulation period	Assumed to be zero

Chart 10. Items calculated in the module for realized gains on securities

Next, (1) gains/losses on sales of bondholdings (excluding credit investments) and stockholdings and (2) the devaluation of bondholdings and stockholdings (excluding credit investments and alternative investments) are calculated. For (1), valuation gains/losses on each type of securities are classified into valuation gains/losses on bondholdings and valuation gains/losses on stockholdings, and it is assumed that sales of these securities during the simulation period are made in line with the average amount of the realized gains/losses on bondholdings and stockholdings over the past three years, which are then regarded as gains/losses on sales of bondholdings and gains/losses on sales of stockholdings.⁷¹ If valuation gains on stockholdings are lower than the average

⁷¹ Many regional banks bolster their net income by selling securities (i.e., realizing gains) as the profitability of their domestic deposit-taking and lending activities continues to decline, and the specification here reflects such behavior by banks. For more details, see, for example, the October 2018 FSR. Meanwhile, regarding the strategic stockholdings of banks, although banks may be sitting on valuation gains, they may not be able to sell them at the same pace as in the past due to the strategic relationships with the firms; however, in the simulations it is assumed that they can be sold in the same way as other types of stocks. Finally, in the FMM it is assumed that banks do not sell credit investments to realize gains.

realized gains/losses on stockholdings over the past three years, it is assumed that the difference is made up for by realized gains on bondholdings (excluding credit investments).⁷² Moreover, when valuation losses on stockholdings arise, it is assumed that devaluations of stockholding are incurred. Specifically, using the estimated loss-cut rule (obtained using nonlinear least squares estimation for the relationship between the market-to-book ratio of stocks and stock investment trusts and the devaluation ratio), it is assumed that impairment losses corresponding to the extent of the decline in the market-to-book ratio are incurred, where separate loss-cut rules are estimated for banks and *shinkin* banks.

Specification of realized gains/losses on securities



⁷² However, to ensure consistency with banks' past behavior, an upper limit is set to the percentage of valuation gains on bondholdings that are realized. Specifically, we identify the period when each individual bank realized the largest amount of gains on bondholdings in the past, and the upper limit is set to the median of the percentages for those periods, which is about 30 percent. Moreover, even if realized gains/losses on bondholdings are lower than the average of the past three years, it is assumed that the difference is not made up for by realized gains/losses on stockholdings.

3.2.3. Credit costs

Credit costs are calculated as the sum of domestic and foreign credit costs for internationally active banks, and as domestic credit costs for domestic banks and *shinkin* banks (Chart 11). Domestic credit costs are divided into those on corporate loans⁷³ and those on loans to individuals, with the latter consisting of credit costs on housing loans and credit costs on the other loans to individuals.⁷⁴ While credit costs on corporate loans and on housing loans are modeled separately,⁷⁵ those on other personal loans are not modeled due to their small size, and the credit cost ratio for other personal loans is assumed to be the same as that for corporate loans.



Note: The \star denotes items that are modeled as endogenous variables. The letters and numbers in square brackets show the correspondence to the income statement items in Chart 6.

Credit cost model for domestic corporate loans

Starting with credit costs for domestic corporate loans, for net loan loss provisions and write-offs, the model is constructed using the transition matrix between different risk categories of borrowers (estimated based on the historical amount of loans for each transition) for each individual bank (Chart 12).⁷⁶ Losses on sales of nonperforming loans

⁷³ Corporate loans include not only loans to nonfinancial corporations but also specialized lending (project finance, object finance, real estate finance, etc.) and loans to the financial industry.

⁷⁴ For example, credit card loans fall into this category.

⁷⁵ In recent years, accounting regimes overseas, particularly in the United States and Europe, have moved from incurred loss provisioning to expected credit loss (ECL) provisioning (in Europe, ECL provisioning based on IFRS 9 was introduced in 2018, while in the United States the Financial Accounting Standards Board introduced current ECL provisioning in 2020). As a result, there may be structural changes in the relationship between macroeconomic variables such as GDP and credit costs recorded by banks. Going forward, if such structural change arises in Japan, we plan to examine how to deal with this in the model while referring to examples overseas.

⁷⁶ The BOJ conducts a questionnaire survey of banks that hold current accounts at the BOJ regarding the classification of their borrowers on a semi-annual basis in the case of banks and on an annual basis

and recoveries of write-offs are assumed to be zero during the simulation period in all scenarios.



Chart 12. Transition between risk categories of borrowers

Banks make loan loss provisions and write-offs when a borrower is downgraded (for example, if the borrower transitions from being a "normal" borrower to one that "needs attention"),⁷⁷ and the loss is calculated by multiplying the loan amount by the applicable loan-loss provision ratio (in the case of loans other than those where the borrower becomes "bankrupt or de facto bankrupt" and losses arise directly).⁷⁸ Moreover, for loans to "bankrupt or de facto bankrupt" borrowers, the loan-loss provision ratio is set to 100 percent and the loss is counted toward credit costs. The credit cost calculation for loans to borrowers that are "in danger of bankruptcy" and to "bankrupt or de facto bankrupt" borrowers also takes into account the percentage of loans covered by collateral or guarantees (the coverage ratio), which varies across banks.⁷⁹ This model structure is in line with the way that banks calculate credit costs in practice.⁸⁰

in the case of *shinkin* banks. Borrower classifications are based on the self-assessment categories specified in the Financial Services Agency's Inspection Manual for Financial Institutions. There are five categories: "normal" borrowers, borrowers that "need attention," borrowers that require "special attention," borrowers that are "in danger of bankruptcy," and borrowers that are "bankrupt or de facto bankrupt." The borrower classification transition matrix data obtained from this survey makes it possible to capture the transition of borrowers and their loans from the beginning of the period to each category at the end of the period.

⁷⁷ The nonperforming loan ratio used in the aforementioned domestic lending interest rate model is defined as the percentage of firms classified as requiring "special attention" or below in each period, as calculated in the borrower classification transition model. The default rate used in the risk-weighted asset model for exposures to corporate loans, discussed below, is defined as the probability that a firm has been downgraded to "special attention" or below during the past six months in this borrower classification transition model.

⁷⁸ Generally, the lower the internal rating, the higher the loan-loss provision ratio.

⁷⁹ The coverage ratio is specified such that it is positively linked with land prices.

 $^{^{80}}$ As noted above, the BOJ's questionnaire survey regarding the classification of borrowers is conducted semi-annually for banks and annually for *shinkin* banks. While the transition probabilities

The transition probabilities between risk categories of borrowers are assumed to be linked to borrower firms' financial conditions and are specified to depend on the GDP growth rate, which is a proxy for borrower firms' sales and profits, and on the ICR, which indicates borrowers' interest payment capacity. The GDP growth rate can be regarded as an indicator of borrower firms' short-term liquidity, while the ICR is an indicator of their medium- to long-term debt repayment capacity. In general, a deterioration in the macroeconomic conditions and/or firms' financial position is seen to have a nonlinear effect on the probability of downgrades and the default rate.⁸¹ For this reason, in the panel estimation of transition probabilities, a logistic model taking such nonlinearities into account is used. Moreover, instead of the economy-wide GDP growth rate and ICR, the weighted averages of industry-level GDP growth rates and ICRs employing the industry composition of individual banks' corporate loans as weights are used. This makes it possible to take into account the heterogeneity in the industry composition of banks' loans.

When calculating the GDP growth rate and ICR for individual banks, the FMM takes into account not only the heterogeneity in the industry composition of loans but also in the composition of borrower firms' size – i.e., whether they are large firms or small and medium-sized enterprises (SMEs) –. In addition, the FMM also considers the heterogeneity among SMEs in terms of whether they are low-return borrowers (i.e., borrowers that, although classified as "normal" borrowers, are in fact in a relatively poor financial condition (so-called medium-risk firms)). The low-return borrowers are defined following the April 2018 FSR (Chapter VI) as firms with a relatively weak financial position (i.e. firms whose ROAs are below the median and whose leverages are above the median of all firms) whose borrowing interest rates are low relative to their credit risk through business cycles, and taking such heterogeneity into account allows for the possibility that the sensitivity of these low-return borrowers to risk factors may differ

between borrower categories therefore are estimated on a semi-annual basis for banks and on an annual basis for *shinkin* banks, in the forward-looking simulations loan-loss provisions and write-offs are calculated on a quarterly basis. For example, to obtain banks' loan loss provisions and write-offs in the April-June quarter of 2022, which is assumed to be the first quarter of the simulation period, we first estimate the values for the semiannual period from January to June 2022. The amounts for the January-March quarter, a rough estimate of which is obtained by dividing the actual values for the semiannual period from October 2021 to March 2022 by two, are then subtracted from the estimate to obtain the values in the April-June quarter. For subsequent quarters, the values are similarly calculated on a semiannual basis and the simulated values for the previous quarter are subtracted.

⁸¹ Using firm-level microdata, the FSR Appendix "A Forecast Model for the Probability of Default Based on Granular Firm-Level Data and Its Application to Stress Testing" (May 2019) shows that (1) firms' probability of default tends to increase nonlinearly as their interest payment capacity decreases, and that (2) such an increase in the probability of default is much larger for more leveraged firms. In addition, it has been shown that, in Japan, the relationship between the ICR and the probability of default is also nonlinear (see the October 2018 FSR for details).
from that of the other borrowers.⁸² The share of SMEs and the share of low-return borrowers among SMEs in banks' loan portfolios are assumed to remain unchanged during the simulation period from the actually observed values just before the simulation period.

For the estimation of domestic credit costs, the estimates of the parameters in linear regression are used for the baseline scenario, while for the stress scenario the estimates of the parameters in 90th percentile quantile regression are used. Thus, by using different parameters corresponding to the economic and financial conditions in the scenarios, the nonlinear relationship between the independent and dependent variables is reproduced in the model.

⁸² When simulating the credit costs of individual banks, we proceed as follows. We first estimate the borrowers' ICR for each borrower category defined in terms of firms' industry, size, and low-return borrower status separately, taking into account the possibility that the sensitivity to macroeconomic fluctuations of each borrower category may differ. We then create an index by taking the weighted average of the ICRs using individual banks' shares of loans to these different borrower categories as weights. Finally, regarding this as a sufficient statistic to represent borrower firms' financial condition, we calculate banks' credit costs. Meanwhile, low-return borrowers are defined as SMEs that meet one of the following two criteria in two consecutive years: (1) the firm's operating ROA is below the median of the distribution of all firms, but its borrowing interest rate is lower than that of the most creditworthy firms in the ROA distribution (i.e., firms in the top 10 percent of the distribution), and/or (2) the firm's financial leverage is above the median of the distribution of all firms, but its borrowing interest rate is lower than that of firms with a relatively high creditworthiness in the financial leverage distribution (i.e., firms in the bottom 50 percent of the distribution). The estimation uses granular firmlevel data, including information on the banks firms transact with. The reason for treating low-return borrowers separately from other firms is that their loans outstanding tend to be large and their ICRs tend to be low, so that their interest payment capacity is likely to be low. As a result, the probability of default of such firms is likely to increase more than that of other firms in the event of a negative macroeconomic shock such as an economic downturn, an increase in procurement costs, and/or an increase in interest rates. In fact, during the global financial crisis, low-return borrowers tended to experience a greater deterioration in their financial indicators (ratio of operating profits to sales and ICR) than non-low-return borrowers. For details, see the April 2018 FSR (Chapter VI). Note that the criteria for determining low-return borrowers are similar to those for so-called zombie firms (see Caballero, Hoshi, and Kashyap 2008, Fukuda and Nakamura 2011, and Kwon et al. 2015), which are defined as firms whose borrowing rates are below the most favorable rates. However, while zombie firms are the result of forbearance lending by banks, low-return borrowers are the result of banks' search for yield, so that the underlying concepts differ somewhat (Kawamoto et al. 2020).

Credit cost model for domestic corporate loans Domestic credit costs (corporate loans)_i = Net loan-loss provisions_i + Write-offs_i = $\Delta \sum_{n=1}^{4} \begin{bmatrix} \text{Exposure (to borrowers in n)}_i \times \text{Provision rate (of n)}_i \end{bmatrix} \\ \times \text{Uncovered ratio (only for } n = 4)_i \end{bmatrix} \\ + \text{Exposure (to borrowers in n = 5)}_i \times \text{Uncovered ratio (for } n = 5_i) \\ \text{Exposure (to borrowers in n)}_i \\ = \sum_{m=1}^{4} \begin{bmatrix} \text{Exposure (to borrowers in m, previous period)}_i \\ \times \text{Probability of transition}_i^{m \to n} \end{bmatrix} \times \text{Domestic loan growth}_i \\ < \text{Transition probability of borrower risk category from m to n for bank } i (PT_i^{m \to n}) > \\ \ln \left(\frac{PT_i^{m \to n}}{1 - PT_i^{m \to n}}\right) = \alpha_1^{m \to n} \times GDP \text{ growth}_i + \alpha_2^{m \to n} \times ICR_i + \text{Fixed effect}_i + \text{Constant} \\ --m \text{ and n denote risk categories of borrowers. } \Delta \text{ denotes difference. } \alpha_1^{m \to n} \text{ and} \\ \alpha_2^{m \to n} \end{array}$

Credit cost model for domestic housing loans

Credit costs for housing loans are calculated by first estimating the rate of loans overdue by three months or more, then calculating the hypothetical credit cost ratio in the absence of credit guarantees, and finally taking into account the portion of loans covered by credit guarantees. To estimate the rate of overdue loans, the unemployment rate to capture the employment and income situation of borrowers and (three-month) government bond yields to represent the interest rate burden are used as explanatory variables. The rate of overdue loans rises with increases in the unemployment rate⁸³ and interest rates. The credit cost ratio in the absence of credit guarantees is assumed to be linked to this rate of overdue loans. The credit cost ratio is then calculated by multiplying the hypothetical credit cost ratio of each bank. Finally, credit costs are calculated by multiplying the credit cost ratio by the outstanding amount of housing loans.

⁸³ The scenario for the unemployment rate assumes that the unemployment rate responds with a certain sensitivity (estimated value) to changes in the domestic output gap. Since the domestic output gap is linked to the level of domestic GDP, given potential GDP, if domestic GDP is treated as an endogenous variable, the unemployment rate is also an endogenous variable.



Credit cost model for foreign loans

The credit cost model for foreign loans is a model using transition probabilities between different risk categories of borrowers and has a similar structure as the credit cost model for domestic corporate loans.⁸⁴ While the growth rate of borrower firms' sales is used as an explanatory variable instead of the GDP growth rate, since the sales growth rate is specified such that it is predicted by the GDP growth rate, etc., the GDP growth rate is still used in effect as an explanatory variable. In addition, the ICR is used in the same form as in the model for domestic corporate loans. The same is true for the growth rate of sales and the ICR for each bank, which are calculated by taking the weighted averages of industry-level growth rate of sales and the ICR using the industry composition of banks' loans to foreign corporations as weights.

However, when estimating the sensitivity of the transition probabilities to the sales growth rate and the ICR, data on the transition probabilities of external ratings in the corporate bond market for foreign firms are used, because long-term time series data for those transition probabilities are available.⁸⁵ Specifically, the sensitivity of the transition probabilities to foreign firms' sales growth rate (at the sample median) and ICR (at the sample median) are estimated, and the transition probabilities for each bank during the simulation period are estimated by adding the estimated change in transition probabilities during the simulation period (due to changes in the sales growth rate, etc.) to the actually observed transition probabilities just before the simulation period. Note that, due to the small sample size in terms of the number of historical transition probabilities used for this

⁸⁴ Losses on nonperforming credit sales and recoveries of write-offs are assumed to be zero during the simulation period in all scenarios.

⁸⁵ Data from Moody's credit risk calculator is used.

estimation, linear regression instead of quantile regression is used to estimate parameters.

Credit cost model for foreign loans

Foreign credit costs $_{i} = \sum_{c=1}^{3} (\text{Net loan} - \text{loss provisions}_{c,i} + \text{Write-offs}_{c,i})$ Net loan-loss provisions_{c,i} + Write-offs_{c,i} $= \Delta \sum_{n=1}^{4} \begin{bmatrix} \text{Exposure (to borrowers in } n)_{c,i} \\ \times \text{Provision rate (of } n)_{c,i} \\ \times \text{Uncovered ratio (only for } n = 4)_{c,i} \end{bmatrix}$ +Exposure (to borrowers in $5)_{c,i} \times \text{Uncovered ratio for } n = 5_{c,i}$ Exposure (to borrowers in $n)_{c,i}$ $= \sum_{m=1}^{4} \begin{bmatrix} \text{Exposure (to borrowers in } m, \text{previous period})_{c,i} \\ \times \text{Probability of transition}_{c,i}^{m \to n} \\ \times \text{Foreign loan growth}_{i} \end{bmatrix}$ <Transition probability from borrower risk category m to n for bank i in country $c(PT_{c,i}^{m \to n}) >$ $ln\left(\frac{PT_{c,i}^{m \to n}}{1 - PT_{c,i}^{m \to n}}\right) = \alpha_{1}^{m \to n} \times GDP \text{ growth}_{c,i} + \alpha_{2}^{m \to n} \times ICR_{c,i} + \text{Fixed effect}_{c,i} + \text{Constant}$ --m and n denote risk categories of borrowers. c denotes foreign countries and regions(America, Europe and Asia and Pacific). Δ denotes difference. $\alpha_{1}^{m \to n}$ and $\alpha_{2}^{m \to n}$ denote the coefficients to predict transition of borrower risk category from m to n.

3.3. Module for valuation gains/losses on securities holdings

The module for valuation gains/losses on securities holdings is used to calculate the valuation gains/losses at the end of each period on securities held by banks. Specifically, valuation gains/losses are determined by adding estimated changes in market value during the period to the valuation gains/losses at the beginning of the period, and then making adjustments based on the sales and devaluation of securities determined in the module for realized gains/losses on securities holdings (Chart 13).⁸⁶ Calculations for stocks etc. are made separately for stocks and investment funds, while calculations for bonds are made separately for government bonds and credit products (Chart 14). The following explains how changes in the market values of these financial instruments during the period are estimated.⁸⁷

⁸⁶ Proceeds from the sale and devaluation of securities are assumed to be reinvested at current market values at the time. Further, reinvestments of an amount equal to the sale or redemption amount (book value) are assumed to be made in a manner that maintains the composition of banks' portfolios at market values.

⁸⁷ The financial variables used to calculate valuation gains/losses on securities holdings are shown in Chart A4 in Appendix 3.



Chart 13. Flow of calculation of valuation gains/losses on securities holdings

Chart 14. Securities in the module for valuation gains/losses on securities holdings

on gains/losses on securities holdings
luation gains/losses on bondholdings
Government bonds, etc
Domestic credit products
Coporate bonds (AAA, AA, A and BBB rated or lower, incl. bank debentures)
Foreign government bonds
Dollar-denominated
Euro-denominated
Foreign credit products
Coporate bonds (AAA, AA, A, BBB, BB, B and CCC rated or lower, incl. bank debentures)
CLO (AAA, AA, A, BBB and BB rated or lower)
CMBS (AAA, AA, A and BBB rated or lower)
RMBS
ABS (AAA and AA rated or lower)
Direct lending funds
Bank loan funds (BB rated or higher and B rated or lower)
ation gains/losses on stockholdings
Domestic stocks and stock investment trusts
Domestic investment funds and alternative investments
Bond investment funds
Credit investment funds
Real estate investment funds
Balanced funds
Hedge funds
Foreign stocks and stock investment trusts
Foreign investment funds and alternative investments
Real estate investment funds
Hedge funds
Private equity

Changes in the market value of JGBs etc. and foreign government bonds during the simulation period are calculated as changes in the discounted present value of future principal and coupon income using the domestic and foreign government bond interest rates by remaining maturity in the scenario. While the way that changes in the market value of domestic and foreign credit products are estimated is generally the same as that for JGBs etc., interest rates of these products are calculated by adding exogenously determined credit risk premiums (spreads) for different types of products and ratings to government bond interest rates. In addition, we assume that a decline in external ratings (in the line with the scenario for the transition probabilities of external ratings) leads to a further increase in spreads.⁸⁸ For domestic and foreign stocks and alternative investments, changes in market value are calculated in line with changes in stock and fund prices in the scenario for the financial variables.⁸⁹

3.4. Risk-weighted assets module

Risk-weighted assets (RWA) consist of the sum of following three elements: RWA for credit risk, RWA for market risk, and RWA for operational risk.⁹⁰ However, what we model in the form of a function are RWA for credit risk items that account for a large share on a value basis – specifically, corporate and sovereign exposures (including securitization exposures), retail exposures, stock exposures, and (among counterparty credit risks) credit value adjustments (CVA) – as well as RWA for operational risk, which make up the largest share of RWA other than RWA for the credit risk items listed above (Chart 15).

⁸⁸ The lower the rating of a bond, the larger the spread tends to be in the event of stress, and the scenarios are set on this basis.

⁸⁹ For foreign government bonds and stocks as well as foreign credit products and foreign alternative investments, it is assumed that foreign exchange rate risks are fully hedged, so that their market values are not linked to exchange rates.

⁹⁰ Credit risk refers to the risk of incurring losses due to defaults by borrowers or issuers of securities, market risk refers to the risk of incurring losses due to securities price fluctuations in trading operations, and operational risk refers to the risk of incurring losses due to operational errors such as clerical errors or fraud by financial institutions' employees or the failure of IT systems.

р .	1			
Rıs	Risk-weighted assets			
	Credit risk			
		Corporate and sovereign exposures (incl. securitization exposures)		
		Loans (\bigstar)		
		Credit products (Coporate bonds, securitized products) (★)		
		The others		
		Retail exposures (★)		
		Stock exposures (★)		
		$CVA(\star)$		
		The others		
	Ma	arket risk		
	Op	perational risk (*)		

Chart 15. Risk-weighted assets module

Note: The \star denotes items that are modeled as endogenous variables.

RWA for credit risk items other than the ones just mentioned and RWA for market risk are specified as remaining unchanged from the actual values just before the simulation period. While the classification of exposures differs between banks using the standardized approach and those using the internal ratings approach, the above classification is used in the FMM for both types of banks, for the sake of simplicity. The following sections explain the models for RWA for corporate and sovereign exposures, retail exposures, stocks exposures, CVA risks, and operational risks.

Risk-weighted asset model for corporate and sovereign exposures

RWA for corporate and sovereign exposures are divided into RWA for corporate loan exposures (such as loans to corporations and financial institutions),⁹¹ exposures to investments in credit products such as corporate bonds and securitized products, and the other exposures (such as sovereign debt) and calculated as follows.

First, RWA for corporate loan exposures are calculated by multiplying risk weights by the amount of corporate loans. The treatment of risk weights differs for banks using the standardized approach and those using the internal ratings approach: in the case of the former, risk weights are assumed to remain unchanged from their recent actual values,⁹² while in the case of the latter they are calculated by multiplying the risk weights specified in accordance with the Basel regulations by a bank-specific level adjustment term

⁹¹ Specialized lending is also included.

⁹² In practice, even for financial institutions using the standardized approach, risk weights may increase due to a decline in external ratings, etc. However, due to data limitations, the FMM assumes that their risk weights remain unchanged from the end of the pre-simulation period.

calculated based on banks' historical risk weights. The risk weights for corporate loans of banks using the internal rating approach are specified such that the risk weight increases in line with increases over the preceding six months in the probability of default on domestic corporate loans calculated in the credit cost model. Next, RWA for exposures to credit products are specified such that risk weights change in response to changes in the external ratings of securities by domestic and foreign issuers (due to changes in GDP growth, ICRs, etc.) in line with the external ratings-based approach,⁹³ which is one approach to calculate the RWA for securitization exposures. The level of computed risk weights is then adjusted using each bank's historical risk weights. In the calculation of exposures to credit products, market values are used for internationally active banks, while book values are used for domestic banks and *shinkin* banks.⁹⁴ The other RWA are assumed to remain unchanged from the end of the pre-simulation period.

Risk-weighted asset model for corporate loan exposures

RWA for corporate loans_i = RW for corporate loans_i × Corporate loans_i RW for corporate loans_i $= \begin{cases} Actual RW for corporate loans_i$ < For banks adopting standardized approach (SA) >Level adjustment term_i × Estimated RW coefficient_i< For banks adopting internal rating based approach (IRB) > $Estimated RW coefficient_i <math>\equiv \Phi\left(\sqrt{\frac{1}{1-\rho_i}} \times \Phi^{-1}(PD_i) + \sqrt{\frac{\rho_i}{1-\rho_i}} \times \Phi^{-1}(0.999)\right) - PD_i$ $\Phi \cdots$ Cummulative distribution function of standard normal distribution, $PD \cdots$ Averaged probability of default, $\rho \cdots$ Correlation coefficient

⁹³ The external ratings-based approach is an approach in which risk weights are applied based on the ratings of eligible rating agencies and the remaining maturity of securitized products. Other approaches are as follows: the internal ratings-based approach, in which the underlying assets of securitized products are assumed to be directly held and risk weights are calculated by substituting the required capital ratio calculated based on the internal ratings approach into a prescribed formula; the standardized approach, in which the underlying assets of securitized products are assumed to be directly held and risk weights are calculated based on the internal ratings approach into a prescribed formula; the standardized approach, in which the underlying assets of securitized products are assumed to be directly held and risk weights are calculated by substituting the required capital ratio calculated based on the standardized approach into the prescribed formula; and the internal assessment approach, in which the internal ratings assigned by a bank are linked to the ratings by eligible rating agencies and risk weights are calculated by applying the external ratings-based approach.

⁹⁴ Book values are used for the exposures of domestic banks and *shinkin* banks for consistency with the FSA's public notice on domestic banks (the same applies below).

Risk-weighted asset model for retail exposures

RWA for retail exposures consist of RWA for exposures to loans to individuals and RWA for the other retail exposures.

Risk-weighted asset model for exposures to loans to individuals RWA for loans to individuals_i = RW for loans to individuals_i × Loans to individuals_i RW for loans to individuals_i = $\begin{cases} \alpha_{1,i} \times \text{Delinquency rate of housing loan (overdue by more than 3-month)_i} \\ + \text{Fixed effect}_i + \text{Constant < For banks adopting IRB >} \end{cases}$

RWA for exposures to loans to individuals are calculated by multiplying the corresponding risk-weights by the amount of domestic loans to individuals. The risk weighting of loans to individuals also differs between banks that use the standardized approach and those that use the internal rating approach: while in the case of the former, risk weights are assumed to remain unchanged from their recent actual values, in the case of the latter they are specified to fluctuate reflecting changes in borrowers' creditworthiness. Specifically, risk weights for loans to individuals are assumed to depend on the rate of loans overdue by three months or more,⁹⁵ and risk weights increase as the rate of loans overdue rises.⁹⁶ RWA for the other retail exposures are assumed to remain unchanged from the end of the pre-simulation period.

Risk-weighted asset model for stock exposures

RWA for stock exposures are calculated by multiplying exposures to stocks, etc., by the risk weights for stocks, etc. While risk weights for all banks are fixed at their most recent actual values in line with the simple risk weight method under the market-based approach, market values are used for the stock exposures of internationally active banks and book

⁹⁵ Since loans to individuals by Japanese banks primarily consist of housing loans, the same rate of loans overdue is used as a proxy for borrowers' creditworthiness in whole retail loans.

⁹⁶ The sensitivity of risk weights to the rate of loans overdue by three months or more is assumed to differ across banks, reflecting the fact that it does differ in practice. Meanwhile, the formula for calculating RWA for retail exposures for banks using the internal rating approach is broadly similar to that used for corporate exposures; however, since data on the probability of default, etc., are not available for retail loans, RWA are modeled using a reduced form equation with the rate of loans overdue by three months or more, which is considered to be correlated with the probability of default, by estimating the sensitivity of risk weights to the rate of loans overdue.

values for those of domestic banks and shinkin banks.97

<internationally active="" banks=""></internationally>
RWA for stocks exposures _i
= Actual RW for stocks, etc. $_i$
× Outstanding amount of stocks, etc. (market value) $_i$
<domestic banks=""></domestic>
RWA for stocks exposures _i
= Actual RW for stocks, etc. $_i$
× Outstanding amount of stocks, etc. (book value) _i

Risk-weighted asset model for stock exposures

Risk-weighted asset model for CVA risk

CVA risk⁹⁸ is modeled only for some large banks with large amounts of RWA for CVA risk, while for the other banks RWA for CVA risk are assumed to remain unchanged from the most recent actual values. For the former, U.S. corporate bond spreads (with a BBB rating) are used as an explanatory variable in order to incorporate changes in average external ratings (probability of default) of counterparties into changes in CVA risk.⁹⁹

⁹⁷ There are three methods for calculating RWA for stock exposures: (1) the simple risk weight method under the market-based approach, in which prescribed fixed risk weights are used; (2) the internal models method, in which financial institutions use internal risk measurement models; and (3) the PD/LGD approach, in which stock exposures are regarded as corporate exposures. Due to data constraints, in the FMM RWA for stock exposures are modeled using the simple risk weight method under the market-based approach for all banks.

⁹⁸ CVA risk refers to the risk of losses in derivative transactions due to a counterparty downgrade. Of the counterparty losses on derivatives that materialized during the 2008 global financial crisis, only one-third were due to actual counterparty defaults, while the remaining two-thirds were due to CVA losses. Therefore, when Basel III was introduced, capital rules were set out that take CVA risk as part of counterparty credit risk into account (Basel Committee on Banking Supervision 2011).

⁹⁹ There are three methods for measuring CVA risk – the advanced risk measurement method, the standard risk measurement method, which is used by many internationally active banks, and the simple method (under which the RWA for CVA risk equal 12 percent of RWA for derivative transactions), which is used by most domestic banks and *shinkin* banks. In the FMM, CVA risk is modeled based on the standard risk measurement method. However, since it is difficult to obtain data such as on the exposure to each counterparty and the maturity and outstanding notional amounts of CVA risk hedging instruments, which are needed for the standard risk measurement method, only changes in CVA risk due to changes in the external credit rating of counterparties are specified and estimated in reduced form. This specification, in which CVA risk changes in line with changes in counterparties' probability of default, is similar to the model used in the Dodd-Frank Act Stress Tests in the United States (Federal Reserve 2021). Meanwhile, since a large share of counterparties (excluding central counterparty clearing houses) in the derivatives transactions of Japanese banks are non-residents, the model uses U.S. corporate bond spreads rather than Japanese ones.

Risk-weighted asset model for CVA risk

RWA for CVA risk _i	
= Actual values of RWA for CVA risk $_i$	
× (1 + α_1 × Δ U.S. corporate credit spread (BBB rated))	

Risk-weighted asset model for operational risk

Regulatory requirements for operational risk allow for the use of the basic indicator approach, the gross income allocation approach and the advanced measurement approach.¹⁰⁰ In the FMM, operational risk is linked to gross operating income (excluding realized gains/losses on bondholdings)¹⁰¹ in a manner that mimics the basic indicator approach for all banks.¹⁰² Since this item represents only a small percentage of each bank's total RWA, a relatively simple specification is used.

Risk-weighted asset model for operational risk

RWA for operational risk_i = $\alpha_1 \times \text{Gross operating income}_i + \text{Fixed effect}_i + \text{Constant}$

3.5. Capital adequacy module

The capital adequacy module consists of the model to calculate the amount of capital that forms the numerator of the regulatory capital adequacy ratio, i.e., Common Equity Tier 1 capital (CET1 capital) in the case of internationally active banks and core capital in the case of domestic banks and *shinkin* banks. In both cases, the model focuses on changes in banks' capital during the simulation period relative to their actual amount of capital just before the simulation period.

¹⁰⁰ In the basic indicator approach, operational risk is set to 15 percent of the average annual gross income over the preceding three years. In the gross income allocation approach, annual gross income is allocated to eight business segments (retail banking, commercial banking, payment and settlement, retail brokerage, trading and sales, corporate finance, agency services, and asset management). The gross income of each segment is then multiplied by a fixed percentage and the three-year average of the resulting sum is taken as the amount of operational risk. In the advanced measurement approach, the amount of operational risk is assumed to be equivalent to the maximum expected operational risk loss calculated based on the operational risk measurement approach employed in the bank's internal controls.

¹⁰¹ Gross operating income = Net interest income + Net fees and commissions + Net trading income + Other net non-interest income.

¹⁰² In the basic indicator approach RWA for operational risk is calculated as the average of 15 percent of gross income (= gross operating income – realized gains/losses on bondholdings) + fees and commissions payments over the most recent three years. However, since fees and commissions payments are not modeled in the FMM, the RWA for operational risk is approximated by gross operating income (excluding realized gains/losses on bondholdings).

In terms of internationally active banks, their CET1 capital is divided into CET1 components and CET1 regulatory adjustments. Major items contributing to changes in CET1 components are net income (after tax) calculated in the periodic profit/loss module, dividends, valuation gains/losses on available-for-sale securities calculated in the module for valuation gains/losses on securities holdings, and foreign currency translation adjustments (Chart 16).

Chart 16. Overview of the capital adequacy module (Internationally active banks)



Specifically, net income (after tax) calculated in the periodic profit/loss module is added to the amount of capital at the beginning of the period, and dividend payments calculated based on the dividend model described below are subtracted. Furthermore, any changes in valuation gains/losses on available-for-sale securities or foreign currency translation adjustment related to foreign subsidiaries, etc., are added or subtracted. When valuation gains increase, for example, deferred tax assets on temporary differences shrink, assuming a uniform effective tax rate of 40 percent.

In addition, CET1 regulatory adjustments are specified such that intangible assets and deferred tax assets on temporary differences, which are assets that are difficult to use to absorb losses, are deducted in line with regulations, while double gearing regulations¹⁰³ are also taken into account (Chart 17).

¹⁰³ Double gearing occurs when one financial institution invests in other financial institutions. Double gearing regulations aim to limit double gearing in order to prevent the propagation of risks within the financial system, such as a chain of financial institution failures, by deducting a certain amount of such investments from banks' capital. In a simplified manner, the FMM takes the specific details of regulations for (1) intentional cross-shareholdings, (2) investments in minority-owned financial institutions (financial institutions in which the investing financial institutions holds voting rights of 10 percent or less), and (3) investments in other financial institutions (financial institutions in which the investing financial institutions in the provide the investing financial institution holds voting rights of more than 10 percent as well as fellow enterprises) into account.

CE	ET1	(C	ommon Equity Tier 1) capital (CET1 components - CET1 regulatory adjustments)	
CET1 components				
Co			mmon share and retained earnings (value at the previous period + Net income - Dividends)	
			Net income after tax	
			Dividends (★)	
		Ac	cumulated other comprehensive income	
			Valuation on available-for-sale securities	
			Foreign currency translation adjustment (\star)	
			The others	
	CE	T1	regulatory adjustments	
		Ce	rtain items for limited recognition (10%/15% thresholds)	
			Deferred tax assets from temporary differences	
			Other certain items	
		Th	e others	

Chart 17. Capital adequacy module (Internationally active banks)

Note: The \star denotes items that are modeled as endogenous variables.

As for domestic banks and *shinkin* banks, the core capital ratio is calculated. While the flow of the calculation is identical to that for the CET1 capital of internationally active banks, there are some differences in the items taken into account. The main difference is that valuation gains/losses on available-for-sale securities are excluded from core capital components (Chart 18). Therefore, with regard to changes in the market value of available-for-sale securities, only realized changes such as gains/losses on sales or impairments have an effect on capital adequacy ratios.



Meanwhile, in addition to modeling core capital regulatory adjustments for domestic banks and *shinkin* banks, transitional arrangements related to capital financing instruments and accumulated other comprehensive income (such as revaluation reserves for land) stipulated only under domestic regulations, are also taken into account (Chart 19).^{104,105}

Core capital (Core capital components - Core capital regulatory adjustments + Transitional adjustments)						
	Cor	e capital components				
		Common share and retained earnings (value at the previous period + Net income - Dividends)				
		Net income after tax				
		Dividends (*)				
		Part of accumulated other comprehensive income				
	Cor	e capital regulatory adjustments				
	Tra	nsitional arrangements for core capital				

Chart 19. Capital adequacy module (Domestic and *shinkin* banks)

Note: The \star denotes items that are modeled as endogenous variables.

Total capital of internationally active bank and capital of domestic banks, which are the numerators of the total capital adequacy and capital adequacy ratios used as explanatory variables in the loan model and the foreign funding interest rate model, are calculated in accordance with Basel III and regulations for domestic financial institutions. Specifically, internationally active banks' total capital is calculated by adding other Tier 1 capital and Tier 2 capital to the aforementioned CET1 capital, where other Tier 1 capital and Tier 2 capital is, in principle, assumed to remain unchanged over the simulation period.¹⁰⁶ For domestic banks and *shinkin* banks, core capital is used as the successor to capital.¹⁰⁷

Dividend model

The dividend model assumes that if the net income of a bank in a particular period is positive, the bank pays dividends in line with the average dividend payout ratio over the past three years (dividend payout ratio = dividends/net income). However, because the payout ratio tends to be extremely high during periods when net income is small, an upper limit of 30 percent is set. Meanwhile, if a bank's net income is negative, it is assumed that it does not pay a dividend. Share buybacks are not taken into account.

¹⁰⁴ Specifically, transitional arrangements for eligible former non-cumulative perpetual preferred stocks, eligible former capital financing instruments, land revaluation differences, non-controlling interests, and capital financing instruments issued through measures related to the strengthening of capital bases by public authorities are taken into account.

¹⁰⁵ As in the case of internationally active banks, foreign currency translation adjustments are also taken into account, but their contribution to changes in core capital is tiny.

¹⁰⁶ General provisions are specified to fluctuate in line with changes in loans.

¹⁰⁷ While the components included in capital and core capital differ slightly, core capital is used because the discrepancies in the time-series data for the two are small.

Dividend model

Common share and retained earnings_i [chg. from previous year]

= Net income (after $tax)_i$

-max[Net income (after tax)_i × min[γ , Average payout ratio in past 3 years_i], 0]

— where γ is set to 0.3 in order to exclude outliers.

Foreign currency translation adjustment model

The foreign currency translation adjustment model assumes that the foreign exchange differences that arise when translating the financial statements of banks' foreign subsidiaries, etc. into yen depend on the net assets of those subsidiaries and the exchange rate (dollar/yen). The exchange rate sensitivity of foreign currency translation adjustments is estimated assuming that it is identical across banks. Meanwhile, for simplicity, the net assets of foreign subsidiaries are assumed to remain unchanged during the simulation period.

Foreign currency translation adjustment model

Foreign currency translation adjustment_i

= Foreign currency translation adjustment_i [previous quarter]

 $+\alpha_1 \times \text{Exchange rates [USD/JPY, q/q chg.]} \times \text{Foreign subsidiaries' net assets}_i$

— Foreign subsidiaries' net assets are assumed constant during the simulation period.

4. Example of Macro Stress Testing

In this section, we demonstrate the detail of the analysis using the FMM and a hypothetical stress scenario in which a crisis equivalent to the global financial crisis of 2008 materializes in the April-June quarter of 2022 (hereinafter referred to as the "financial stress scenario"), where the end of pre-simulation period is July-September 2021.¹⁰⁸ As mentioned earlier, from the perspective of identifying the risk characteristics of banks, it is useful to examine the deviation of the simulation results in the financial stress scenario from the baseline scenario. We show such comparison in the latter part of this section.

¹⁰⁸ The analytical example presented in this section corresponds to the one analyzed and published as the macro stress test under the "Financial Stress Scenario" in the April 2022 FSR.

4.1. Scenario

The baseline scenario assumes that the real economy follows the average forecasts by professional forecasters as of March 2022 (Chart 20). In terms of financial variables, we assume that asset prices in the market as of mid-March 2022 reflect forecasts of the domestic and foreign economies by market participants. Furthermore, government bond interest rates follow the forward rates factored into the yield curve as of the same date, and stock prices, exchange rates, and domestic and foreign credit spreads remain unchanged from their March 2022 levels (Chart 21).





Source: BEA; Cabinet Office; Eurostat; Haver Analytics; IMF; Japan Center for Economic Research.



Source: Bloomberg; FRB; Haver Analytics.

In the financial stress scenario, we consider a situation in which international financial market experiences a significant deterioration comparable to the global financial crisis in the April-June quarter of 2022, which has a negative impact on financial intermediation and exerts downward pressure on the real economy at home and foreign, causing a feedback loop between domestic economy sector and domestic banking sector.¹⁰⁹

We assume that the paths of foreign economies after April-June 2022 follows a trajectory similar to that observed during the global financial crisis (Chart 20). For the Japanese real economy, we conduct a simulation to estimate the effects of an exogenous shock similar to the global financial crisis on the Japanese economy.

For most financial variables, we assume that a large and rapid adjustment comparable to the global financial crisis occurs in the international financial markets between April and June 2022 (Chart 21).¹¹⁰ However, for some financial variables, we take into account the possibility that the fluctuations in financial variables during times of stress may have changed in recent years, as investment funds have increased their presence in the global financial system. More specifically, we assume that spreads on relatively highly rated bonds are more likely to widen than they did during the global financial crisis, as evidenced by the rapid market downturn in March 2022.¹¹¹ We assume that government bond interest rates fall to record lows and remain unchanged thereafter.

4.2. Simulation results¹¹²

This sub-section illustrates the simulation results in line with the description of the model

¹⁰⁹ At the time of the global financial crisis, the capital adequacy ratios of Japanese banks were relatively low compared to the current level, which may have had a strong feedback loop of restraining lending behavior of banks and further depressing the real economy. Since the crisis, Japanese banks have increased their capital, and assuming the same level of stress in the real economy, the current feedback loop is likely to be relatively muted. In this regard, when constructing scenarios, the paths of foreign real economies and financial variables are exogenously given, and then domestic real GDP is calculated endogenously from the model by incorporating the feedback loop. See, for example, the April 2021 FSR for a discussion of the effect of the accumulation of capital by Japanese banks after the global financial crisis on GDP through the suppression of feedback loop under stress.

¹¹⁰ We assume that the dollar funding premium for all Japanese banks also expand to the same level as during the global financial crisis. The simulation estimates the increase in foreign currency funding costs for each bank, taking into account the shift of each bank's funding source towards deposits after the global financial crisis.

¹¹¹ For U.S. corporate bonds and securitized products, we assume a situation where the increase in the spread for high-rated bonds is larger than that during the global financial crisis. To do this, we first take the ratio of the spread for low-rated bonds during the global financial crisis to that during the March 2020 market turmoil, which is about three for U.S. corporate bonds. We then multiply the ratio with the spread increase for each rating in March 2020 to obtain the degree of increase in the spread during the simulation period.

¹¹² The simulation takes into account the effects of policy responses by the government and the BOJ

structure in Section 3. First, we look at loan outstanding, which is the only factor that fluctuates among the investment and funding accounts. In the baseline scenario, the loan outstanding for both internationally active banks and domestic banks excluding shinkin banks maintains a positive year-on-year growth rate (Chart 22). The loan outstanding for shinkin banks temporarily decreases during fiscal 2022 and 2023, but recovers to positive growth in 2024. The temporary decrease is due to the assumption on the repayment of effectively interest-free loans, which were introduced by the government as a funding support measure for small and medium-sized enterprises (SMEs) after the spread of Covid-19 in 2020 and which contributed to large growth in the loan outstanding for shinkin banks in fiscal 2020.¹¹³ Next, in the financial stress scenario, the loan outstanding for all types of banks is significantly lower than in the baseline scenario, and it declines in fiscal 2022 compared to the previous year. The decline is partly due to the supply side effect, in addition to the demand side effect such that the demand for loans from corporates declines. Specifically, in the financial stress scenario, the capital adequacy ratios of banks, especially internationally active banks, significantly decline to the level close to that required by the capital buffer regulation, suppressing the loan supply of banks. A feedback loop, mentioned in the previous subsection, denotes such phenomenon.



as well as lending by banks since the outbreak of Covid-19 in 2020 through February 2022. For more details, see the macro stress test in the April 2022 FSR. Also, see Appendix 5 for specific implementation of the support measures into the simulation using the FMM.

¹¹³ After the spread of Covid-19, effectively interest-free loans by private banks have been introduced from May 2020 to March 2021 as a measure to support liquidity of SMEs. As a result, the loan outstanding to domestic firms increased in fiscal 2020, especially at regional banks, which tend to lend to SMEs.

Second, we break down the simulation results for periodic profits/losses into three components: PPNR excluding trading income, realized gains/losses on securities holdings, and credit costs. In the baseline scenario, on the one hand, PPNR excluding trading income increases for internationally active banks as their loans outstanding grow steadily, especially to foreign borrowers, while their lending margins do not change significantly (Chart 23). On the other hand, even as loans outstanding for domestic banks and *shinkin* banks continue to increase, their PPNR excluding trading income remains flat due to the impact of a shrinking trend in lending margins caused by fiercer competition in the loan market.¹¹⁴ In contrast, in the financial stress scenario, loans outstanding of internationally active banks decline due to a sharp drop in demand for loans as the real economy deteriorates.¹¹⁵ At the same time, the squeeze in foreign lending margins due to higher foreign currency funding costs contributes to a large decline in PPNR excluding trading income. Due to a slowdown in the domestic economy, domestic banks and *shinkin* banks experience a decrease in loan demand, resulting in a decline in their loans outstanding, although they are relatively unaffected by the rise in foreign currency funding costs.





¹¹⁴ This shrink in lending margin reflects a gradual decline of the loan demand index, an explanatory variable in the domestic loan interest rate model, in line with historical trend.

¹¹⁵ The sensitivity of the foreign loans to changes in the foreign GDP is higher than that of the domestic loans to changes in the domestic GDP. This is based on the average sensitivity during past recessions, including the global financial crisis when foreign lending declined considerably. However, the model may not be able to capture the impact of the increased importance of foreign operations by Japanese banks in recent years.

Realized gains/losses on securities holdings (cumulative values for fiscal 2021-2024) in the baseline scenario are positive due to the realization of gains on stockholdings, while valuation gains/losses on securities holdings just before the simulation period are generally positive in all types of banks (Chart 24). In contrast, in the financial stress scenario, realized gains/losses on securities holdings exhibit loss by large impairment losses on stocks, funds, and credit products due to the shocks to financial variables. In the meantime, the negative impact is partially offset by realized gains on bondholdings, mainly among banks that continue to post positive valuation gains on the holdings of bonds excluding credit products.¹¹⁶





Note: Ratio relative to RWA as of the end of fiscal 2020.

Credit costs in the baseline scenario remain almost unchanged from the period just before the simulation period. In the financial stress scenario, credit costs sharply increase due to the impact of downturn in domestic and foreign economies (Chart 25). Moreover, the credit cost ratios peak in fiscal 2022, when the real economy is most depressed. For internationally active banks with a large share of foreign loans among their portfolio,

¹¹⁶ The increase in realized gains from bonds in the financial stress scenario relative to the baseline scenario is due not only to the mechanism by which losses from stocks are offset by gains from bonds, but also to the fact that valuation gains on bonds increase during the simulation period as government bond interest rates decline.

foreign credit costs substantially contribute to the overall credit cost ratios (Chart 26).¹¹⁷



Chart 25. Credit cost ratios

Chart 26. Credit cost ratios (cumulative totals of fiscal 2021-24)



Third, with regard to valuation gains/losses on securities holdings (change from fiscal 2020 to 2024), in the baseline scenario, valuation gains/losses on bondholdings, excluding credit products, decline in all types of bank, as government bond interest rates rise

¹¹⁷ The reason why the dynamics of the credit cost ratio of banks and that of *shinkin* banks are slightly different is that the frequency of banks' credit cost models is semiannual, while that of *shinkin* banks' models is annual. For instance, when calculating semi-annual and annual changes based on the same quarterly GDP scenario, annual GDP changes are less sensitive to short-term GDP fluctuations. As a result, the fluctuation in the estimates of the credit cost ratio based on annual model tends to be less volatile than that based on semi-annual model.

moderately (Chart 27). In contrast, in the financial stress scenario, valuation gains/losses deteriorate significantly with contributions from the decrease in the values of stocks, fund investments, and credit investments. It should be noted that such valuation gains/losses on securities holdings affect the capital adequacy ratio only for internationally active banks.



Chart 27. Valuation gains/losses on securities holdings (change from fiscal 2020 to 2024)

Fourth, with regard to risk-weighted assets, in the baseline scenario internationally active banks, whose loans outstanding exhibit relatively fast growth, shows a large increase. Conversely, risk-weighted assets of domestic banks excluding *shinkin* banks show a moderate increase, and those of *shinkin* banks remain almost unchanged (Chart 28). In the financial stress scenario, the risk-weighted assets of internationally active banks and domestic banks excluding *shinkin* banks are, on the one hand, pushed up by the increase in the risk weights due to the deterioration of their portfolio, mainly among banks using the internal rating approach. On the other hand, the decrease in loans outstanding push down their risk-weighted assets. The former effect outweighs the latter, so their risk-weighted assets of *shinkin* banks in the financial stress scenario are similar to those in the baseline scenario, as risk-weights remain unchanged due to the use of the standardized approach, while the degree of decrease of loans outstanding is limited.

Note: The estimates take tax effects into account. Ratio relative to RWA as of the end of fiscal 2020.





Based on the aforementioned simulation results in each module, the capital adequacy ratio at the end of fiscal 2024 in the baseline scenario deteriorates somewhat from the fiscal 2020 level (Chart 29), as the increase in PPNR excluding trading income is offset by the increase in credit costs and risk-weighted assets, as well as a deterioration in valuation gains/losses on securities holdings for internationally active banks. In the financial stress scenario, PPNR excluding trading income, realized gains/losses on securities holdings, and credit costs all contribute to downward pressure relative to the baseline scenario, and the capital adequacy ratio at the end of fiscal 2024 is significantly lower than in the baseline scenario (Chart 30). In particular, the capital adequacy ratios of internationally active banks, where valuation gains/losses on securities holdings decline and riskweighted assets increase due to higher risk weight, are significantly depressed. The capital adequacy ratios of some banks fall below the level required by their capital buffer regulation.



Chart 29. Decomposition of capital adequacy ratio (baseline scenario)

Note: 1. The charts indicate the contribution of each factor to the difference between the capital adequacy ratios as of end-March 2021 and the end of the simulation period (as of end-March 2025) under the baseline scenario.2. The left-hand chart shows the CET1 capital ratio of internationally active banks. The middle and right-hand charts show the core capital ratio of domestic banks.

Chart 30. Decomposition of capital adequacy ratio (financial stress scenario)



Note: 1. The charts indicate the contribution of each factor to the difference between the capital adequacy ratios as of the end of the simulation period (end-March 2025) under the baseline and the financial stress scenario.

2. The left-hand chart shows the CET1 capital ratio of internationally active banks. The middle and right-hand charts show the core capital ratio of domestic banks.

So far, we explain the simulation results of the baseline scenario and the financial stress scenario, with a particular emphasis on the contribution of each module described in Section 3 to the difference in the capital adequacy ratios of the two scenarios. Instead, here we consider the contribution of each sector (domestic economy sector, foreign economies sector, and domestic and foreign financial markets sector) (Chart 31).¹¹⁸ First, the CET1 ratio for internationally active banks, which is 11.9% at the end of fiscal 2024 in the baseline scenario, declines to 7.3% in the financial stress scenario. The decline is attributed to the deterioration of the domestic economy sector by -1.3 percentage points, the deterioration of the foreign economies sector by -0.8 percentage points, and the deterioration of the domestic and foreign financial markets sectors by -2.5 percentage points.¹¹⁹ The deterioration of the domestic and foreign economies substantially affects the capital adequacy ratios mainly through higher credit costs resulting from the deterioration of borrowers' creditworthiness. The deterioration in the domestic and foreign financial markets affects the capital adequacy ratios through the deterioration in realized and valuation gains/losses on securities holdings, which results from the decline in the market value of securities holdings, and through the decrease in PPNR excluding trading income, which results from higher funding interest rates and lower interest and dividend income. Next, for domestic banks excluding shinkin banks, the capital adequacy ratio, which is 9.8% at the end of fiscal 2024 in the baseline scenario, declines to 7.4% in the financial stress scenario. The decline is attributable to the deterioration of the domestic economy by -1.3 percentage points and the deterioration of domestic and foreign financial markets by -1.1 percentage points.¹²⁰ The deterioration in the domestic economy, as in the case of internationally active banks, affects the capital adequacy ratios of domestic banks mainly through an increase in the credit costs. The deterioration in domestic and foreign financial markets affects the capital adequacy ratios through a deterioration in realized gains/losses on securities holdings, which results from a decline in the market value of securities holdings, and through PPNR excluding trading income, which results from a decrease in interest and dividend income. Unlike internationally active banks, the

¹¹⁸ Specifically, for example, the contribution of the domestic economy sector is calculated as the difference between the capital adequacy ratio in the baseline scenario and that obtained by running a simulation based on the scenario variables (exogenous variables), among which those belonging to the domestic economy sector are equal to the values in the financial stress scenario and those belonging to other sectors are equal to the values in the baseline scenario.

¹¹⁹ The impact of a nonlinear term and other factors are included in the deterioration of the domestic and foreign financial markets sector. The impact of the deterioration in domestic and foreign financial markets alone is -2.5 percentage points.

¹²⁰ The impact of a nonlinear term and other factors are included in the deterioration of the domestic and foreign financial markets sector. The impact of the deterioration in domestic and foreign financial markets alone is -0.9 percentage points.

domestic banks excluding *shinkin* banks are largely unaffected directly by the deterioration in the foreign economies sector due to the relatively small size of their foreign business. In terms of *shinkin* banks, the capital adequacy ratio, which is 12.4% at the end of fiscal 2024 in the baseline scenario, declines to 10.1% in the financial stress scenario. This is because of a -0.5 percentage point contribution from the deterioration in the domestic economy sector and a -1.7 percentage point contribution from the deterioration in the domestic and foreign financial markets sector. ¹²¹ While the qualitative features of the contributions of each sector are similar to those of the domestic banks excluding *shinkin* banks, their relatively high ratio of credit investments to risk-weighted assets makes *shinkin* banks more susceptible to deterioration in the domestic and foreign financial markets sector.

Chart 31. Decomposition of capital adequacy ratio by sector (financial stress scenario)



Note: 1. The charts indicate the contribution of each sector to the difference between the capital adequacy ratios as of the end of the simulation period (end-March 2025) under the baseline scenario and the financial stress scenario.

2. The left-hand chart shows the CET1 capital ratio of internationally active banks. The middle and right-hand charts show the core capital ratio of domestic banks.

Relatedly, to demonstrate the nonlinearity in the simulation results of the FMM against the degree of stress on the scenario variables, Appendix 4 provides simulation results in the "1/2 financial stress scenario," which assumes that the degree of stress on each scenario variable is equal to a half of that in the financial stress scenario.

¹²¹ The impact of a nonlinear term and other factors are included in the deterioration of the domestic and foreign financial markets sector. The impact of the deterioration in domestic and foreign financial markets alone is -1.6 percentage points.

5. Conclusion

The Financial Macro-econometric Model (FMM) employed by the BOJ is a model constructed to simulate the financial conditions of the domestic banking sector in the event of stress and is used mainly for the assessment of risks to the financial system overall, including in the macro stress tests in the FSR.

This paper described the framework of the FMM as of September 2022, focusing on the specification of the domestic banking sector. Compared to the original model in 2011, when the Bank of Japan started using the FMM, the coverage of the model has been substantially expanded, so that the current version of the FMM is able to describe endogenous changes in banks' balance sheets and profits and losses in the event of stress, and also includes the modeling of risk-weighted assets, dividends, and other capital adequacy variables. Moreover, the current FMM also incorporates a variety of complex transmission mechanisms of shocks to the economy. For example, an increase in credit costs in the FMM leads to both a deterioration in periodic profits/losses and an increase in risk-weighted assets, which lower capital adequacy ratios, which then reduces the supply of loans. In addition, lower capital adequacy ratios result in an increase in foreign currency funding costs due to a decline in banks' own creditworthiness.

However, it should be noted that even with these improvements, the FMM still represents only an approximation of certain aspects of the actual financial system and the real economy. Although the FMM has been developed to well explain the developments in historical data and ensure a certain degree of consistency with economic theory, like all other economic models, it cannot perfectly capture economic and financial developments. Moreover, even if the model itself represents a more or less accurate description of reality, quantitative assessments based on the model are subject to technical limitations. For instance, the estimation of parameters is based on historical average relationships between macroeconomic and financial variables and the parameters may not sufficiently incorporate structural changes that have occurred recently. In addition, some of the estimates are based on the assumption that the parameters related to financial variables are the same for all banks of a particular type, and thus may not sufficiently incorporate the heterogeneity across individual banks.

The followings are some points regarding the direction of future improvements to the FMM. The first point concerns work to continue deepening the understanding of the risk characteristics of various exposures held by banks and reflect them in the model. For example, while the model for credit risks related to corporate loans in the FMM is designed to reflect differences in the risk characteristics of individual borrowers to a

reasonable degree, it is important to continue to refine the model in this regard, because of the borrowers' heterogenous risk characteristics, as shown by the differences in the variation in corporate profits across industries and firms observed in the wake of outbreak of the Covid-19 pandemic in 2020 and by the differences in risk characteristics in the real estate industry highlighted in the April 2022 edition of the Financial System Report. In addition, using granular data, it would be desirable to analyze the impact of firm-level shocks on the banking sector and the economy at the macro level¹²² and the transmission of shocks through transaction relationships among firms and banks¹²³ and to reflect these in the model.

The second point concerns the refinement of the modeling of the behavior of economic and financial variables during crises, including liquidity crises. Since the FMM currently focuses on sufficiency of banks' capital adequacy, liquidity risks are not explicitly considered. However, once a fire sale, for example, occurs – i.e., a situation arises in which debt-related stress at a particular financial institution results in the deterioration of the securities portfolios of many financial institutions through the dumping of the value of assets – the scale of stress may increase. From the perspective of assessing the resilience of the financial system to stress, it is necessary to deepen our understanding of the phenomena that occur at times of crisis but are rarely observed during normal times, for example by using high-frequency data, and incorporate them into the model.

The third point concerns the expansion of coverage. From the perspective of expanding the coverage of the range of risks faced by individual financial institutions, it is important to model financial institutions' total exposure, including off-balance sheet items, and subsidiaries. Moreover, in terms of risks faced by the financial system, it is also important to appropriately take the role of non-bank financial institutions, including liquidity risks in the non-bank sector, into account in the model. Another issue to be considered in the future is incorporating the impact of the implementation of the Basel III Finalization, such as in the risk-weighted asset models.¹²⁴ Going forward, it will be essential to continue improving the FMM in order to appropriately respond to new issues arising from changes in the environments surrounding Japan's financial system.

¹²² See Arata and Miyakawa (2021, 2022) for details.

¹²³ See, for example, Fukunaga and Miyakawa (2022).

¹²⁴ Specifically, the Basel III Finalization stipulates revisions to the standardized and internal ratings approaches for credit risk, CVA risk measurement methods, and measurement methods for market risk and operational risk, as well as the introduction of a capital floor (see Basel Committee on Banking Supervision 2017b).

Reference

Acharya, V., R. Engle, and S. Steffen [2021] "Why Did Bank Stocks Crash During COVID-19?" NBER Working Papers No 28559.

Adrian, T., N. Boyarchenko, and D. Giannone [2019] "Vulnerable Growth," *American Economic Review*, 109 (4), pp. 1263-1289.

Adrian, T., F. Grinberg, N. Liang, S. Malik, and J. Yu [2022] "The Term Structure of Growth-at-Risk," *American Economic Journal: Macroeconomics*, 14(3), pp 283-323.

Adrian, T., J. Morsink and L. Schumacher [2020] "Stress Testing at the IMF," Departmental Paper Series (Monetary and Capital Markets Department), No.20/04, February 2020.

Afonso, G., A. Kovner, and A. Schoar [2011] "Stressed, Not Frozen: The Federal Funds Market in the Financial Crisis," *Journal of Finance*, 66(4), pp. 1109-1139.

Aikman, D., P. Alessandri, B. Eklund, P. Gai, S. Kapadia, E. Martin, N. Mora, G. Sterne, and M. Willison [2011] "Funding Liquidity Risk in a Quantitative Model of Systemic Stability," Central Banking, Analysis, and Economic Policies Book Series, in: Rodrigo Alfaro (ed.), *Financial Stability, Monetary Policy, and Central Banking*, 1(15-12), pp. 371-410, Central Bank of Chile.

Alessandri, P., P. Gai, S. Kapadia, N. Mora, and C. Puhr [2009] "Towards a Framework for Quantifying Systemic Stability," *International Journal of Central Banking*, 5(3), pp. 47-81.

Amiti, M., and D. Weinstein [2018] "How Much Do Idiosyncratic Bank Shocks Affect Investment? Evidence from Matched Bank-Firm Loan Data" *Journal of Political Economy*, 126(2), pp. 525-587.

Anderson, R., J. Danielsson, C. Baba, U. S. Das, H. Kang, and M. Segoviano [2018] "Macroprudential Stress Tests and Policies: Searching for Robust and Implementable Frameworks," IMF Working Paper, WP/18/197, September 2018.

Aoki, R., K. Antoku, S. Fukushima, T. Yagi and S. Watanabe [2021] "Foreign Currency Funding of Major Japanese Banks - Review of the March 2020 market turmoil -," Bank of Japan Review Series, No.21-E-4, October 2021.

Arata, Y., and D. Miyakawa [2021] "The Size of Micro-originated Aggregate Fluctuations: An analysis of firm-level input-output linkages in Japan," Discussion papers 21066, Research Institute of Economy, Trade and Industry (RIETI).

Arata, Y., and D. Miyakawa [2022] "Demand Shock Propagation Through an Input-output Network in Japan," Discussion papers 22027, Research Institute of Economy, Trade and Industry (RIETI).

Baba, N., and F. Packer [2009] "Interpreting Deviations from Covered Interest Parity during the Financial Market Turmoil of 2007-08," *Journal of Banking & Finance*, 33(11), pp. 1953-1962.

Bank of Japan [2011] "The Bank of Japan's Initiatives on the Macroprudential Front," Bank of Japan, October 2011, https://www.boj.or.jp/en/finsys/fs_policy/fin111018a.pdf.

Basel Committee on Banking Supervision (BCBS) [2009] "Principles for Sound Stress-Testing Practices and Supervision." May 2009.

Basel Committee on Banking Supervision (BCBS) [2011] "Capital Treatment for Bilateral Counterparty Credit Risk Finalised by the Basel Committee" June 2011.

Basel Committee on Banking Supervision (BCBS) [2013] "Liquidity Stress Testing: a Survey of Theory, Empirics and Current Industry and Supervisory Practices" Working Paper

No. 24, October 2013.

Basel Committee on Banking Supervision (BCBS) [2017a] "Supervisory and Bank Stress Testing: Range of Practices" December 2017.

Basel Committee on Banking Supervision (BCBS) [2017b] "Basel III: Finalising Post-Crisis Reforms" December 2017.

Basel Committee on Banking Supervision (BCBS) [2021] "The Basel Framework"

Baudino, P., R. Goetschmann, J. Henry, K. Taniguchi and W. Zhu [2018] "Stress-Testing Banks – A Comparative Analysis," FSI Insights on policy implementation, No.12, November 2018.

Bernanke, B., M. Gertler, and S. Gilchrist [1996] "The Financial Accelerator and the Flight to Quality," *The Review of Economics and Statistics*, 78(1), pp. 1-15.

Berrospide, J., and R. Edge [2010] "The Effects of Bank Capital on Lending: What Do We Know, and What Does It Mean?" *International Journal of Central Banking*, 6(34), pp. 1-50.

Berrospide, J., R. Meisenzahl, and B. Sullivan [2012] "Credit Line Use and Availability in the Financial Crisis: the Importance of Hedging," Finance and Economics Discussion Series 2012-27, Board of Governors of the Federal Reserve System (U.S.).

Board of Governors of the Federal Reserve System (FRB) [2019] "Amendments to Policy Statement on the Scenario Design Framework for Stress Testing (12 CFR Part 252)," Federal Register, 84(40), February 28, 2019.

Board of Governors of the Federal Reserve System (FRB) [2021] "Dodd-Frank Act Stress Test 2021: Supervisory Stress Test Methodology," Dodd-Frank Act Stress Test Publications, April 2021.

Brayton, F., T. Laubach., and D. Reifschneider [2014] "The FRB/US Model: A Tool for Macroeconomic Policy Analysis," FEDS Notes. Washington: Board of Governors of the Federal Reserve System, April 03, 2014.

Budnik, K., M. Balatti, I. Dimitrov, J. Groß, I. Hansen, G. di lasio, M. Kleemann, F. Sanna, A. Sarychev, N. Sinenko and M. Volk [2019] "Macroprudential Stress Test of the Euro Area Banking System," ECB Occasional Paper, No.226, July 2019.

Budnik, K., M. Balatti, I. Dimitrov, J. Groß, M. Kleemann, T. Reichenbachas, F. Sanna, A. Sarychev, N. Siņenko, and M. Volk [2020] "Banking Euro Area Stress Test model," Working Paper Series No. 2469, European Central Bank.

Bundesbank [2015] "Macro stress tests – Technical Documentation," 28 July 2015.

Caballero, R., T. Hoshi, and A. Kashyap [2008] "Zombie Lending and Depressed Restructuring in Japan," *American Economic Review*, 98 (5), pp. 1943-1977.

Camara, B., F. Castellani, H. Fraisse, L. Frey, C. Héam, L. Labonne, V. Martin [2015] "MERCURE: A Macroprudential Stress Testing Model developed at the ACPR," Débats économiques et financiers 19, Banque de France.

Chernenko, S., and A. Sunderam [2014] "Frictions in Shadow Banking: Evidence from the Lending Behavior of Money Market Mutual Funds," *The Review of Financial Studies*, 27(6), pp. 1717-1750.

Copeland, A., A. Martin, M. Walker [2014] "Repo Runs: Evidence from the Tri-Party Repo Market," *Journal of Finance*, 69(6), pp. 2343-2380.

Correia, S., K. Kiernan., M. Seay, and C. Vojtech [2020] "Primer on the Forward-Looking Analysis of Risk Events (FLARE) Model: A Top-Down Stress Test Model," Finance and Economics Discussion Series 2020-015, Board of Governors of the Federal Reserve System (U.S.).

Correia, S., M. Seay, and C. Vojtech [2022] "Updated Primer on the Forward-Looking Analysis of Risk Events (FLARE) Model: A Top-Down Stress Test Model," Finance and Economics Discussion Series 2022-009, Board of Governors of the Federal Reserve System (U.S.).

Covitz, D., N. Liang, and G. Suarez [2013] "The Evolution of a Financial Crisis: Collapse of the Asset-Backed Commercial Paper Market," *Journal of Finance*, 68(3), pp. 815-848.

Dees, S., J. Henry, and R. Martin [2017] "Stamp€: Stress-Test Analytics for Macroprudential Purposes in the Euro Area," European Central Bank, February 2017.

Dent, K., B. Westwood, and M. Segoviano [2016] "Stress Testing of Banks: an introduction," Bank of England Quarterly Bulletin, 56(3), pp. 130-143.

European Banking Authority (EBA) [2021] "2021 EU-Wide Stress Test: Methodological Note," 13 November 2020.

European Central Bank (ECB) [2021a] "Financial Stability Review," November 2021, European Central Bank.

European Central Bank (ECB) [2021b] "Macroprudential Stress Test of the Euro Area Banking System amid the Coronavirus (COVID-19) Pandemic," October 2021, European Central Bank.

European Systemic Risk Board (ESRB) [2020] "Macro-Financial Scenario for the 2020 EU-Wide Banking Sector Stress Test," ESRB Letter on 2020 EU-wide Stress Test, January 2020.

Fazzari, S., G. Hubbard, and B. Peterson [1988] "Financing Constrains and Corporate Investment," *Brookings Papers on Economic Activity*, 1988(1), Brookings Institution, pp. 141-206.

Fender, I., M. Gibson, and P. Mosser [2001] "An International Survey of Stress Tests," Current Issues in Economics and Finance, Federal Reserve Bank of New York, 7(10).

Financial System and Bank Examination Department of the Bank of Japan, and Strategy Development and Management Bureau, and Supervision Bureau of the Financial Services Agency [2020] "Supervisory Simultaneous Stress Testing Based on Common Scenarios," Bank of Japan Review Series, No. 20-E-9, December 2020.

Fique, J. [2017] "The MacroFinancial Risk Assessment Framework, Version 2.0," Bank of Canada Technical Report No. 111.

Fukuda, S., and J. Nakamura [2011] "Why Did 'Zombie' Firms Recover in Japan?" *The World Economy*, 34 (7), pp. 1124–1137.

Fukunaga, I., N. Hara, S. Kojima, Y. Ueno, and S. Yoneyama [2011] "The Quarterly Japanese Economic Model (Q-JEM): 2011 version," Bank of Japan Working Paper Series, No. 11-E-11, November 2011.

Fukunaga, K, and D. Miyakawa [2022] "Supply Chain Network and Credit Supply," IMES Discussion Paper Series 22-E-08, Institute for Monetary and Economic Studies, Bank of Japan.

Henry, J., and C. Kok [2013] "A Macro Stress Testing Framework for Assessing Systemic Risks in the Banking Sector," Occasional Paper Series 152, European Central Bank.

Hirakata, N., K. Kan, A. Kanafuji, Y. Kido, Y. Kishaba, T. Murakoshi, and T. Shinohara [2019] "The Quarterly Japanese Economic Model (Q-JEM): 2019 version," Bank of Japan Working Paper Series, 19-E-7, June 2019.

Hirtle, B., A. Kovner, J. Vickery, and M. Bhanot [2016] "Assessing Financial Stability:

The Capital and Loss Assessment under Stress Scenarios (CLASS) Model," *Journal of Banking and Finance*, 69(S1), pp S35-S55.

Ichiue, H., T. Kitamura, S. Kojima, T. Shirota, K. Nakamura and N. Hara [2009] "Hybrid Model of Japanese Economy: Quarterly-Japanese Economic Model (Q-JEM), " Bank of Japan Working Paper Series, No. 09-J-6, July 2009 (in Japanese).

Ishikawa, A., K. Kamada, Y. Kurachi, K. Nasu and Y. Teranishi [2012] "Introduction to the Financial Macro-econometric Model," Bank of Japan Working Paper Series, No. 12-E-1, January 2012.

Kamada, K. and Y. Kurachi [2012] "Impact of the Changes in Government Bond Interest Rates on Financial Sectors and Economy - Analysis Based on Financial Econometric Model - ," RIETI Discussion Paper Series, 12-J-021, July 2012 (in Japanese).

Kawamoto, T., T. Matsuda, K. Takahashi, and Y. Tamanyu [2020] "Bank Risk Taking and Financial Stability: Evidence from Japan's Loan Market," Bank of Japan Working Paper Series, No.20-E-1, February 2020.

Kawata, H., T. Kitamura, K. Nakamura, Y. Teranishi, and S. Tsuchiya [2012] "Effects of the Loss and Correction of a Reference Rate on Japan's Economy and Financial System: Analysis Using the Financial Macro-econometric Model," Bank of Japan Working Paper Series, No.12-E-11, December 2012.

Kawata, H., Y. Kurachi, K. Nakamura and Y. Teranishi [2013] "Impact of Macroprudential Policy Measures on Economic Dynamics: Simulation Using a Financial Macro-econometric Model," Bank of Japan Working Paper Series, No. 13-E-3, February 2013.

Kitamura, T., S. Kojima, K. Nakamura, K. Takahashi and I. Takei [2014] "Macro Stress Testing at the Bank of Japan," Bank of Japan Research Papers, October 2014.

Kitamura, T., I. Muto, and I. Takei [2016] "Loan Interest Rate Pass-Through and Changes after the Financial Crisis: Japan's Evidence," *Journal of the Japanese and International Economies*, 42, pp. 10-30.

Kiyotaki, N., and J. Moore [1997] "Credit Cycles," *Journal of Political Economy*, 105(2), pp. 211-248.

Klein, M. [1971] "A Theory of the Banking Firm", *Journal of Money, Credit, and Banking*, 3(2-1), pp. 205-218.

Kwon, H., F. Narita, and M. Narita [2015] "Resource Reallocation and Zombie Lending in Japan in the 1990s," *Review of Economic Dynamics*, 18 (4), pp. 709-732.

Krishnamurthy, A., S. Nagel, and D. Orlov [2014] "Sizing Up Repo," *Journal of Finance*, 69(6), pp. 2381-2417.

Liang, N. [2018] "Well-Designed Stress Test Scenarios Are Important for Financial Stability," Blog post at the Brookings Institution, February 2018

Lopez-Salido, D., J. Stein, and E. Zakrajsek [2017] "Credit-Market Sentiment and the Business Cycle," *The Quarterly Journal of Economics*, 132(3), pp.1373-1426.

Mancini, L., A. Ranaldo, and J. Wrampelmeyer [2016] "The Euro Interbank Repo Market," *The Review of Financial Studies*, 29(7) pp. 1747-1779.

Monti, M. [1972] "Deposit, Credit, and Interest Rate Determination under Alternative Bank Objectives" in *Mathematical Methods in Investment and Finance*, G.P. Szego and K. Shell (eds.), Amsterdam: North-Holland.

Nagahara, T., and T. Sekine [2002] "The Effects of Monetary Policy on Firm Investment after the Collapse of the Asset Price Bubble: An Investigation Using Japanese Micro Data," Bank of Japan Working Paper Series, No. 02-E-3, July 2002.

Oliver, B., D. Learmonth, J. McKeown, and R. Williams [2012] "RAMSI: a top-down stress-testing model developed at the Bank of England," Bank of England Quarterly Bulletin, Bank of England, 52(3), pp. 204-212.

Ozaki, M., T. Konno, H. Hiroyama and S. Tsuchiya [2019] "Cross-Prefecture Lending by Regional Banks," Bank of Japan Review Series, No. 2019-J-4, May 2019 (in Japanese).

Peek, J., and E. Rosengren [1997] "The International Transmission of Financial Shocks: The Case of Japan," *American Economic Review*, 87(4), pp. 495-505.

Peek, J., and E. Rosengren [2000] "Collateral Damage: Effects of the Japanese Bank Crisis on Real Activity in the United States," *American Economic Review*, 90(1), pp. 30-45.

van Leuvensteijn, M., C. Kok, J. Bikker, A. van Rixtel [2013] "Impact of Bank Competition on the Interest Rate Pass-Through in the Euro Area," *Applied Economics*, 45(11), pp. 1359-1380.

van Leuvensteijn, M., J. Bikker, A. van Rixtel, and C. Kok [2011] "A New Approach to Measuring Competition in the Loan Markets of the Euro Area," *Applied Economics*, 43(23), pp. 3155-3167.

Appendix 1: Structure of the Model of the Domestic Economy Sector

This Appendix shows how the domestic economy sector in the FMM responds to external stress (Chart A1). The stress for domestic economy in stress scenarios of macro stress testing is often derived from changes in economic and financial conditions outside domestic economy, such as the downturn in foreign economies and substantial adjustment in international financial market. In such scenarios, we in principle estimate the stress for banks by using the estimates on the endogenous response of domestic economy sector, represented by each component of GDP, due to the external stress.

The FMM predicts year-on-year rate of changes in GDP and its components using yearon-year rate of changes in explanatory variables. This is in contrast to typical macroeconomic models used to predict future developments of GDP, in which GDP is estimated by monthly or quarterly changes in variables seasonally adjusted by programs such as X-12-ARIMA. The contrast is due to the fact that many modules in the FMM include variables from financial statements of banks, to which it is not easy to reliably apply seasonal adjustments; we therefore use year-on-year rate of changes to remove seasonality.

Noi	ninal GDP (Household expenditure + Private non-residential investment
	+ Private inventory + Government expenditure + Net export)
	Nominal household expenditure (\bigstar)
	Nominal private non-residential investment (\bigstar)
	Nominal changes in private inventory
	Nominal government expenditure
	Nominal net export (Nominal export - Nominal import)
	Nominal export (\bigstar)
	Nominal import (\bigstar)
GD	P (Household expenditure + Private non-residential investment
	+Private inventory+Government expenditure+Net export)
	Household expenditure
	Private non-residential investment
	Changes in private inventory
	Government expenditure
	Net export (Export—Import)
	Export (\bigstar)
	Import (\bigstar)

Chart A1. Models for domestic GDP

Note: The \star denotes items that are modeled as endogenous variables.

Nominal GDP is composed of nominal household expenditure (the sum of private consumption and private residential investment), nominal private non-residential

investment, changes in nominal private inventories, nominal government expenditure, nominal exports and nominal imports. Among these components, the FMM simulates the movement of nominal household expenditure, nominal non-residential investment, nominal exports and nominal imports using the economic and financial conditions assumed in a scenario. In contrast, the FMM assumes that changes in nominal private inventories remain zero, because their large fluctuations are mainly caused by factors that are independent of macroeconomic developments. The FMM further assumes that nominal government expenditure remains unchanged from the actual value at the end of the pre-simulation period.

Most of the components of real GDP are calculated by dividing the corresponding component of nominal GDP by the deflator, which is exogenously set based on the forecasts by professional forecasters. The exceptions are the exports and imports, whose nominal and real values are estimated separately, because the fluctuations in real exports and imports are often smaller than those in nominal exports and imports in time of stress, such as the global financial crisis (GFC).¹²⁵

Model for nominal household expenditure

The explanatory variables for nominal household expenditure include nominal compensation of employees and stock prices.¹²⁶ The former is regarded as a proxy of disposable income, while the latter is a proxy of the value of assets held by households. In addition, as some of household expenditure is funded by debt, the explanatory variables include the amount of loans to individuals and domestic lending interest rates. These variables work as the channel through which the stress on domestic banks influence real economy. For example, when the tightening of credit supply decreases loans to individuals and increases domestic lending interest rates, the household expenditure in the model decreases due to the severer borrowing constraints for households.

¹²⁵ An alternative way would be to construct the models for nominal export/import and export/import deflator, and obtain real export/import by dividing the nominal values by the corresponding deflator. However, we could not establish suitable models that satisfactorily explain the time series data of export/import deflator by macroeconomic variables included in the FMM. We therefore ended up adopting the models that directly explain real export/import.

¹²⁶ Nominal compensation of employees (year-on-year rate of changes) are determined by nominal GDP (year-on-year rate of changes) and labor share (change from previous year). The labor share during the simulation period is exogenously determined.

Specification of the nominal household expenditure model

Household expenditure [y/y chg.]

= $\alpha_1 \times$ Nominal employee compensation [y/y chg.] + $\alpha_2 \times$ stock prices [y/y chg.] + $\alpha_3 \times$ household loans [y/y chg.]

 $+\alpha_4 \times \text{Domestic lending interest rate [chg. from previous year]} + \text{Constant}$

Model for nominal non-residential investment

The explanatory variables for nominal non-residential investment include the expected growth rate of the domestic economy for the coming three years, which is a proxy of expected returns of the investment.¹²⁷ In addition, as some firms face large costs of external funding and some are even subject to borrowing constraints, the explanatory variables of the model include corporate profitability, the amount of corporate loans, and domestic lending interest rates.¹²⁸ The latter two variables work as the channel through which the stress on domestic banks influence real economy. For example, when the tightening of credit supply decreases corporate loans and increases domestic lending interest rates, the non-residential investment in the model decreases due to the severer financial constraints for corporates.

Specification of the nominal non-residential investment model

Private investment [y/y chg.]
$= \alpha_1 \times \text{Expected}$ economic growth rate
$+\alpha_2 \times \text{Domestic lending interest rate [chg.from previous year]}$
$+\alpha_3 \times \text{Domestic corporate loans [y/y chg.]}$
$+\alpha_4 \times \text{Domestic corporate profit ROA [chg. from previous year] + Constant}$

Model for nominal exports

It is assumed that nominal exports primarily depend on the GDP for the countries or regions to which Japan mainly exports and the nominal effective exchange rate. Here, the GDP represents the demand from the foreign countries or regions, while the nominal effective exchange rate reflects the relative prices of the goods exported from Japan. As for the GDP, we multiply it by the contemporaneous shares of the countries or regions

 $^{^{127}\,}$ The expected growth rate of the macro economy, in turn, depends on the growth rate of potential GDP.

¹²⁸ The increase in the internal funding for firms, represented by higher profitability, may lead to the increase in investment, given that the cost of obtaining external funding is more expensive than the cost associated with internal funding. This is empirically confirmed by, for example, Fazzari, Hubbard, and Peterson [1988] and Nagahata and Sekine [2002].
among the whole export destinations in order to take into account the gradual shift of the destination from the United States and Europe to Asia and Pacific over the estimation period.¹²⁹ Note that the GDP of each destination shows strongly positive correlation with the nominal exports at the following events: the burst of the IT bubble in the early 2000s and the GFC for the United States; the GFC and European sovereign debt crisis in the early 2010s for Europe; and Asian financial crisis in the late 1990s and the GFC for Asia and Pacific.

In addition, the explanatory variables for nominal exports include the oil price, which is regarded as a proxy of commodity prices, whose change influences the cost of producing the goods to be exported and thus leads to fluctuations in the export deflator. Furthermore, the four-quarter lag of nominal exports are also included to remove technical and deterministic fluctuations due to the fact that the export function features year-on-year rate of changes of variables.

Specification of the nominal export model					
Nominal exports [y/y chg.]					
$= \alpha_1 \times U.S.GDP$ growth [y/y chg.] × Export share of U.S.					
$+\alpha_2 \times EU$ GDP growth [y/y chg.] × Export share of EU					
$+\alpha_3 \times Asia$ and Pacific GDP growth [y/y chg.] \times Export share of Asia and Pacific					
$+\alpha_4 \times \text{Nominal effective exchange rate [y/y chg.]}$					
$+\alpha_5 \times \text{Crude oil prices [y/y chg.]}$					
$+\alpha_6 \times \text{Nominal exports [4-quarter lag, y/y chg.]} + \text{Constant}$					

Model for nominal imports

It is assumed that nominal imports depend on the nominal effective exchange rate just as the nominal exports do. In addition, the explanatory variables for nominal imports include the oil price, which captures the fluctuations in commodity price, and the nominal exports, which capture the contribution of the imports of intermediate goods. Note that while it is in principle reasonable to include some proxies of domestic demand such as household consumption and GDP of Japan as the explanatory variables of imports, we do not do that in order to decrease the number of variables simultaneously determined during the simulation and thus reduce computational complexity. The four-quarter lag of nominal imports are included to remove technical and deterministic fluctuations due to the fact that the import function, just as the export function, features year-on-year rate of changes

¹²⁹ The export share of each country/region during the simulation period is assumed to remain the same as the actual value at the end of the pre-simulation.

in variables.

Specification of the nominal import model

Nominal imports [y/y chg.]

= $\alpha_1 \times$ Crude oil prices [y/y chg.] + $\alpha_2 \times$ Nominal effective exchange rate [y/y chg.]

 $+\alpha_3 \times \text{Nominal exports [1-quarter lag, y/y chg.]}$

 $+\alpha_4 \times \text{Nominal imports [4-quarter lag, y/y chg.]} + \text{Constant}$

Model for real exports

On the one hand, the function for real exports is similar to that of nominal exports in that it depends on the GDP for the destination countries or regions, representing the demand from the destination. The explanatory variables also include the four-quarter lag of the dependent variable. On the other hand, the explanatory variables include not nominal effective exchange rate but real effective exchange rate as a proxy for the competitiveness of Japanese firms in international markets. They do not include oil price either.

Specification of the real export model

Real exports [y/y chg.]= $\alpha_1 \times U.S.GDP$ growth $[y/y \text{ chg.}] \times Export$ share of U.S. + $\alpha_2 \times EU$ GDP growth $[y/y \text{ chg.}] \times Export$ share of EU + $\alpha_3 \times Asia$ and Pacific GDP grwoth $[y/y \text{ chg.}] \times Export$ share of Asia and Pacific + $\alpha_4 \times Real$ effective exchange rate [y/y chg.]+ $\alpha_5 \times Real$ exports [4-quarter lag, y/y chg.] + Constant

Model for real imports

The set of explanatory variables for real imports is different from that for nominal imports in that nominal effective exchange rate is replaced by real effective exchange rate and that the oil price, reflecting the fluctuation of import deflator, is removed.

Specification of the real import model

Real imports [y/y chg.]= $\alpha_1 \times \text{Real effective exchange rate } [y/y \text{ chg.}]$ + $\alpha_2 \times \text{Real exports } [y/y \text{ chg.}]$ + $\alpha_3 \times \text{Real imports } [4-\text{quarter lag, } y/y \text{ chg.}] + \text{Constant}$

Appendix 2: Structure of the Model of the Domestic and Foreign Corporate Sectors

The FMM includes system of equations that explains variables in the financial statements of corporates using macroeconomic variables. Important subset of the FMM models such as the model for credit costs is based on those variables and equations.

The models of domestic corporate sector include equations for corporates by industry, size, and profitability (Chart A2). The classification by industry is composed of four industries: (1) processing industries, including food products and beverages, fabricated metal products, general-purpose, production and business oriented machinery, electrical machinery, equipment and supplies, transport equipment, and some parts of other manufacturing; (2) services for individuals, including accommodation and food services, living-related services, amusement, learning support, medical, health care and welfare; (3) transport and postal services; and (4) the others. The classification by size has two categories, large corporates and SMEs, while the classification by profitability also has two categories, low return borrowers and the other corporates. The growth rate of GDP (change from half a year ago, a proxy of the growth rate of sales) and the interest coverage ratio (hereafter ICR) are adopted as the variables that capture creditworthiness of corporates of each industry. The growth rate of GDP is regarded as a proxy for the short-term liquidity, while the ICR is regarded as a proxy for medium-term solvency.



Chart A2. Summary of domestic economy sector

The ICR is calculated by dividing ROA by the product of leverage ratio and borrowing interest rate. Both the ROA and the leverage ratio depend on the GDP, or more specifically the output gap for the corresponding sector.¹³⁰ The borrowing interest rate is assumed to be linked with the lending interest rate for domestic banking sector.

Specification of the models of domestic corporate sector						
Operating profits _j + Interest and dividends received _j						
Interest payment _j						
_ Operating ROA _j						
$\frac{1}{1}$ Borrowing interest rate _j × Leverage ratio _j						
$ROA_j = \alpha_1 \times \text{Output gap}_j + \alpha_2 \times \text{Import price index} + \alpha_3 \times \text{Exchange rates [USD/JPY]}$						
+Fixed $effect_j$ + Constant						
Leverage ratios _j [chg. from previous year] = $\alpha_1 \times \text{Output gap}_j$ [chg. from previous year]						
Borrowing interest rate _j [chg. from previous year]						
= Domestic lending interest rate [chg.from previous year]						
— <i>j</i> denotes industry.						

The models of foreign corporate sector include equations for corporates by region and industry (Chart A3). The classification is based on three regions, America, Europe and Asia-Pacific, and on five industries, (1) processing industries, (2) services for individuals, (3) transport, (4) energy, and (5) the others.¹³¹

¹³⁰ As for the output gap, historical values are based on the estimates by the Bank of Japan, while the values during the simulation period are estimated using Hodrick-Prescott filter.

¹³¹ Unlike the domestic corporate sector, the foreign corporates are not classified by size. As is the case for the domestic corporate sector, the ICR and the growth rate of sales, both of which are the explanatory variables for foreign credit costs, are calculated as the weighted average using each industry's share among total loans outstanding for the bank as a weight.



Chart A3. Summary of foreign economies sector

The models of foreign corporate sector are primarily explained by the GDP of the corresponding region and industry.¹³² Specifically, we adopt the growth rate of sales (change from half a year ago) and ICR as the variables that represent the creditworthiness of the firms belonging to each region and industry. As in the models of domestic corporate sector, the growth rate of sales is regarded as a proxy of borrowers' liquidity, and the ICR is regarded as a proxy of borrowers' solvency. In addition, for the energy industry, the sales and ICR are assumed to depend on oil prices as well. The formulation of ICR is slightly different from that for the domestic corporate sector, as it is explained by output gap of the region, the leverage ratio which depends on the output gap as well, and borrowing interest rates.

The borrowing interest rates of America and Asia and Pacific are assumed to depend on 10-year Treasury bond yield, while the borrowing interest rates of Europe are assumed to depend on 10-year German bond yield. For Asia and Pacific, we simply use the U.S. interest rates, as the interest rates in this region were to some extent correlated with the U.S. interest rates in the past.

¹³² The financial variables for firms among energy sector are assumed to depend on the oil prices, represented by WTI.

Specification of the models of foreign corporate sector Sales growth rate_{c,j} = $\alpha_{1,c} \times \text{Real GDP growth rate}_{c,j}$ [h/h chg.] + $\alpha_{2,c,j} \times \text{GDP deflator}_{c,j}$ [h/h chg.] + $\alpha_{3,c,\text{Energy industry}} \times \text{Crude oil prices}$ +Fixed effect_{c,j} + Constant $ICR_{c,j} = \alpha_{1,c} \times \text{Output gap}_c + \alpha_{2,c,j} \times \text{Leverage ratios}_{c,j} + \alpha_{3,c} \times \text{Borrowing interest rate}_c$ + $\alpha_{4,c,\text{Energy industry}} \times \text{Crude oil prices} + \text{Fixed effect}_{c,j} + \text{Constant}$ Leverage ratios_{c,j}[chg.from previous year] $= \alpha_{1,c} \times \text{Output gap}_c$ [chg.from previous year] — j denotes industry, c denotes foreign countries and regions.

Appendix 3: Variables of Domestic and Foreign Financial Markets

For some of the variables for domestic and foreign financial markets, the path during simulation period is exogenously assumed for each scenario, often based on the historical developments of the variables (Chart A4). Such variables include those that are regarded as quantitatively important risk factors that significantly contribute to the movement of market prices of securities held by Japanese banks. The variables also include those that significantly influence consumption, investment and/or external trade of households and firms.

Dom	nestic financial market							
	Bonds							
	Yields on government bonds (3 month - 40 years)							
	Credit products							
	Corporate credit spread (AAA, AA, A, BBB or lower)							
	Stocks							
	Stock price (TOPIX)							
	Investment funds and alternative investments							
	Bond investment fund's price							
Credit investment fund's price								
Real estate investment fund's price								
Balanced fund's price								
Hedge fund's price								
	The others							
	Exchange rates (USD/JPY)							
Fore	ign financial market							
	Bonds							
Yields on U.S. government bonds (3 month - 30 years)								
Yields on Germany government bonds (3 month - 30 years)								
	Credit products							
U.S. corporate credit spread (AAA, AA, A, BBB, BB, B, CCC or lo								
CLO spread (AAA, AA, A, BBB, BB or lower)								
	CMBS spread (AAA, AA, A, BBB or lower)							
	RMBS spread							
	ABS spread (AAA, AA or lower)							
	Direct lending fund's price							
	Bank loan fund's price (BB, B or lower)							
	Stocks							
	U.S. Stocks (S&P500)							
	Investment funds and alternative investments							
Real estate investment fund's price								
	Hedge fund's price							
	Private equity investment fund's price							
	The others							
	Oil price (WTI)							
Dollar interbank offered rate								
	Dollar funding premiums							

Chart A4. List of financial market variables

Note: Option-adjusted spreads are used for credit products in the foreign financial market.

Appendix 4: Simulation Results of "1/2 Financial Stress Scenario"

This appendix shows the simulation results of "1/2 financial stress scenario," which assumes a half of the stress of "financial stress scenario" examined in Section 4, and compares the results of "1/2 financial stress scenario" with those of "financial stress scenario."¹³³ This comparison enables us to grasp the degree of nonlinearity of the FMM somewhat comprehensively.

For internationally active banks, CET1 ratio as of the end of the simulation period (fiscal 2024) for the financial stress scenario (7.3 %) is lower than that for baseline scenario (11.9%) by 4.6 percentage points. On the other hand, the decrease in CET1 ratio for the 1/2 financial stress scenario (9.9 %) is by 2.0 percentage points, which is smaller than a half of the corresponding figure for the financial stress scenario (Chart 30 and A5). Among the components of the CET1 ratio, significant nonlinearity exists for credit costs, PPNR excluding trading income, realized gains/losses on securities holdings, and changes in risk-weighted assets. The nonlinearity in the credit costs and changes in risk-weighted assets is primarily due to the fact that, to calculate credit costs and risk weights for loans, logistic function is adopted for the model of transition probabilities between different borrower categories. There is nonlinearity in realized gains/losses in securities holdings because realized losses in stocks and credit product investments arise only when the market price substantially falls and banks need to realize impairment losses. As for the PPNR excluding trading income, the nonlinearity is related to the assumption of the FMM that, facing significant realized losses in stockholdings, banks sell bondholdings to realize the gains in order to make up the losses. After selling bondholdings, it is assumed that banks repurchase bonds whose income gains are typically low because the financial stress scenario assumes the lowest interest rates in the history, thus leading to the lowering of PPNR. Since the realized losses in stockholdings are due to impairment losses, so are nonlinear as mentioned above, the lowering of PPNR is also nonlinear.

For domestic banks excluding *shinkin* banks, the core capital ratio as of the end of fiscal 2024 for the financial stress scenario (7.4 %) is lower than that for baseline scenario (9.8 %) by 2.4 percentage points. Conversely, the decrease in the core capital ratio for the 1/2 financial stress scenario (8.9 %) is by 0.9 percentage points, which is again smaller than a half of the corresponding figure for the financial stress scenario. Among the components of the core capital ratio, credit costs and realized gains/losses on securities

¹³³ Specifically, the path of each economic and financial variables in the 1/2 financial stress scenario is set to be equal to the simple average of the corresponding variable in the financial stress scenario and that in the baseline scenario.

holdings primarily drive the nonlinearity.

For *shinkin* banks, due to the similar factors as the domestic banks excluding *shinkin* banks, the core capital ratio as of the end of fiscal 2024 for the 1/2 financial stress scenario (11.5 %) is lower than that for baseline scenario (12.4 %) by only 0.9 percentage points, compared with 2.3 percentage-point decrease for the financial stress scenario (10.1 %).

Finally, we compare each sector's contribution to capital adequacy ratio in the 1/2 financial stress scenario with that in the financial stress scenario (Chart 31 and A6). For all types of banks, the result of the comparison is consistent with the nonlinearity explained above for each component of capital adequacy ratio. Namely, for internationally active banks, significant nonlinearity arises primarily due to domestic and foreign economies sectors, while, for domestic banks including *shinkin* banks, the domestic economy sector and domestic and foreign financial markets sector primarily contribute to pushing down the core capital ratio in a nonlinear manner.



Chart 30. Decomposition of capital adequacy ratio (financial stress scenario)

Note: 1. The charts indicate the contribution of each factor to the difference between the capital adequacy ratios as of the end of the simulation period (end-March 2025) under the baseline scenario and the financial stress scenario.

2. The left-hand chart shows the CET1 capital ratio of internationally active banks. The middle and right-hand charts show the core capital ratio of domestic banks.



Chart A5. Decomposition of capital adequacy ratio (1/2 financial stress scenario)

- Note: 1. The charts indicate the contribution of each factor to the difference between the capital adequacy ratios as of the end of the simulation period (end-March 2025) under the baseline scenario and the 1/2 financial stress scenario.
 - 2. The left-hand chart shows the CET1 capital ratio of internationally active banks. The middle and righthand charts show the core capital ratio of domestic banks.



Chart 31. Decomposition of capital adequacy ratio by sector (financial stress scenario)

- Note: 1. The charts indicate the contribution of each sector to the difference between the capital adequacy ratios as of the end of the simulation period (end-March 2025) under the baseline scenario and the financial stress scenario.
 - 2. The left-hand chart shows the CET1 capital ratio of internationally active banks. The middle and right-hand charts show the core capital ratio of domestic banks.

Chart A6. Decomposition of capital adequacy ratio by sector (1/2 financial stress scenario)



Note: 1. The charts indicate the contribution of each factor to the difference between the capital adequacy ratios as of the end of the simulation period (end-March 2025) under the baseline scenario and the 1/2 financial stress scenario.

2. The left-hand chart shows the CET1 capital ratio of internationally active banks. The middle and righthand charts show the core capital ratio of domestic banks.

Appendix 5: Simulation under the Pandemic

During the period of economic downturn due to the spread of Covid-19 since March 2020, various corporate financing support measures were implemented, including effectively interest-free loans and cash payments by the government. By modifying various aspects of the FMM to incorporate these measures, the macro stress testing shown in the issues of the Financial System Report (FSR) between October 2020 and April 2022 takes into account the effect of such policy measures on the health of banks through helping borrower firms' liquidity and solvency.

This section outlines those modifications of the model regarding the corporate financing support measures, and then examines how accurately the modified FMM can predict the changes in capital adequacy ratio of banks during that period. Specifically, by inputting the FMM both the financial statements of banks in FY 2019 as a starting point of the simulation and the set of actual values of macroeconomic and financial variables during FY 2020 as a "scenario," we calculate the developments in capital adequacy ratio during FY 2020. We then compare the calculated values to the actual values of capital adequacy ratio. Note that the actual values of macroeconomic and financial variables are qualitatively different from those variables assumed for the baseline scenario in macro stress testing shown in FSR.¹³⁴

The spread of Covid-19 is generally considered as the major tail event since the global financial crisis, while the parameters of the FMM are estimated based on historical time series data mostly including the global financial crisis period. Therefore, the simulation results based on the "scenario" of the pandemic is important as a natural out-of-sample tail-event simulation. One should, however, keep in mind that the characteristics of the pandemic recession is qualitatively different from the global financial crisis because the stress from the pandemic originates not from financial markets but from the real economy.

A.5.1. Models for domestic corporate financing support measures¹³⁵

The corporate financing support measures implemented in Japan since the spread of

¹³⁴ The baseline scenario for the macro stress testing shown in FSR assumes that real economy and interest rates follow the forecasts by professional forecasters and the market participants, respectively. It further assumes that the prices of risky assets remain the same as the actual value just before the simulation period. The assumed paths of those variables are clearly different from the actual values during the corresponding period.

¹³⁵ The simulation by the FMM does not take into account the effect of corporate financing support measures by foreign governments. This is partly due to the practical difficulty in comprehensively examining and considering foreign policies. Note also that foreign loans by Japanese banks are concentrated on large firms, which are less likely to be supported by governments than smaller firms.

Covid-19 in fiscal 2020 are primarily composed of cash payments by the government and effectively interest-free loans, both of which aim at supporting SMEs. The cash payments by the government represent various subsidies to firms, such as "subsidies for sustaining businesses," "rent assistance subsidies," and "expansion of employment adjustment subsidies program."¹³⁶ Under certain assumptions, the total amount actually transferred to firms during fiscal 2020 is estimated to be about 1.5 trillion yen for large and medium-sized firms and about 9 trillion yen for smaller firms.¹³⁷ As for the effectively interest-free loans, we consider those loans by private and government-affiliated financial institutions that are accompanied with a transfer to make up the interest payments and guaranteed by the credit guarantee corporations.

The simulation by the FMM takes into account the effect of the cash payments and effectively interest-free loans. Note that the volume outstanding of the effectively interest-free loans, as estimated based on the data provided by credit guarantee corporations, is about 21 trillion yen, of which about 3 trillion yen are by internationally active banks, about 10 trillion yen by domestic banks excluding *shinkin* banks, and about 9 trillion yen by *shinkin* banks. Those amount to 1 percent, 6 percent, and 15 percent of total corporate loans outstanding by each bank type. Thus, it was domestic banks that significantly increased lending as a result of the corporate financing support measures. This was probably due to the fact that SMEs occupy major part of the lending by regional banks. In addition, it was also due to the fact that the face-to-face services industries, which experienced especially severe shortage in demand during the spread of Covid-19, also occupy a large portion of their lending.¹³⁸

The specific method of considering each policy measure in the FMM is the following. As

¹³⁶ In addition to these measures, various other measures were implemented by the government, including the tax payment moratorium, which allowed firms to defer payments of national and local taxes and/or social insurance contributions for one year. However, the simulation in this Appendix only considers cash payments and effectively interest-free loans, because of their large impact.

¹³⁷ As for the details of this estimation, we follow the methods shown in Section A of Chapter IV "Domestic credit risk" from October 2020 issue, April 2021 issue, and October 2021 issue of the FSR.

¹³⁸ This can be seen from the "loans and bills discounted by sector," the official statistics compiled and published by the BOJ. The statistics indicate that the share of face-to-face services industries among total loans is higher for *shinkin* banks than other banks. Note that the face-to-face services industries include hotels and accommodations, food services, living-related services and amusement, all of which were significantly impacted by the spread of Covid-19. In addition, loans outstanding by the size of borrower firms indicate that the amount of loans outstanding to SMEs in face-to-face services industries is larger than that to large firms in the same industries. This implies that, since most of borrowers for major banks are large firms, they are less exposed to face-to-face services industries than regional banks. Note that, since most domestic banks are regional banks, we use the two words interchangeably.

for the effectively interest-free loans, we first divide domestic corporate loans by each bank into the effectively interest-free loans and the other loans. Then we assume that the effectively interest-free loans incur no risk-weighted assets and no credit costs.

Risk-weighted asset model with financing support measures

Risk-weighted assets for corporate loans_i

- = Risk weights for corporate loans_{*i*} × Corporate loans^{*}_{*i*}
 - ---- * denotes loans excluding effectively interest-free loans.

Moreover, both the cash payment and the effectively interest-free loans are assumed to support corporate finance of borrower firms and thus reduce credit costs that are accompanied with "the other loans." Specifically, the credit cost model in the FMM adopts the growth rate of GDP as a proxy of borrowers' short-term liquidity and the ICR as a proxy of borrowers' medium- to long-term solvency. Here, the effectively interest-free loans lower the contribution of the decline in the growth rate to the historically average level. In addition, as for the cash payments, we simply set the ROA of firms higher by the amount comparable to the amount of the cash payments throughout fiscal 2020, which automatically pushes up ICR.¹³⁹

¹³⁹ The methods to take into account corporate financing support measures outlined in this Appendix follow those adopted in the macro stress testing shown in the October 2020 issue, the April 2021 issue, the October 2021 issue and the April 2022 issue of the FSR.



Net loan-loss provisions_i + Write-offs_i $= \Delta \sum_{n=1}^{4} \begin{bmatrix} \text{Exposure (to borrowers in n)}_{i}^{*} \times \text{Provision rate (for n)}_{i} \\ \times \text{Uncovered ratio (only for } n = 4)_{i} \end{bmatrix}$ +[Exposure (to borrowers in 5)_{i}^{*} \times \text{Uncovered ratio (for } n = 5)_{i}] Exposure (to borrowers in n)_{i} $= \sum_{m=1}^{4} \begin{bmatrix} \text{Exposure (to borrowers in m in previous period)}_{i}^{*} \\ \times \text{Probability of transition}_{i}^{m \to n} \end{bmatrix}$ × Domestic loan growth_i = sindicates exposures excluding effectively interest-free loans.<Probability of transition from m to n for bank i (PT_{i}^{m \to n}) > $\ln \left(\frac{PT_{i}^{m \to n}}{1 - PT_{i}^{m \to n}}\right) = \alpha_{1}^{m \to n} \times GDP \text{ growth}_{i}(\text{Average as of before pandemic})$ + Fixed effect_i + Constant $-m \text{ and } n \text{ denote risk categories of borrowers. } \Delta \text{ denotes difference. } \alpha_{1}^{m \to n}$ and $\alpha_{2}^{m \to n}$ denote the coefficients to predict transition from m to n.

A.5.2. Simulation results

The followings show the results of simulation based on the aforementioned scenario and the assumptions on corporate financing support measures (Chart A7). By comparing the simulation results of capital adequacy ratio as of the end of fiscal 2020 with the actual values, we observe that they are quantitatively similar, suggesting that the simulation by the FMM captures the reality rather accurately.



Chart A7. Factors behind changes in capital adequacy ratio (simulation)



2. The left-hand chart shows the CET1 capital ratio of internationally active banks. The middle and righthand charts show the core capital ratio of domestic banks.

Chart A8. Factors behind changes in capital adequacy ratio (actual)



Note: 1. The charts indicate the contribution of each factor to the difference between the actual capital adequacy ratios at end-March 2020 and at end-March 2021.

2. The left-hand chart shows the CET1 capital ratio of internationally active banks. The middle and righthand charts show the core capital ratio of domestic banks.

Central Bank	Model			Feedback loops	
(legal region)	Abbreviation	Official name	Official name Scope of the model	btw the bank lending	Reference
				and the real economy	
EDD		Supervisory stress test models	Financial holding companies covered in Dodd- Frank Act Stress Test	No	FRB [2021]
(US)	FLARE	Forward-Looking Analysis of Risk Events model	All financial holding companies (200 largest financial institutions in asset size are modeled	One way loop from the economic fluctuation	Correia et al. [2020], Correia, Seay, and
			individually and the others are modeled altogether)	to the bank lending	Vojtech [2022]
NY Fed (US)	CLASS	Capital and Loss Assessment under Stress Scenarios model	All financial holding companies (200 largest financial institutions in asset size are modeled individually and the others are modeled altogether)	No	Hirtle et al. [2016]
ECB	BEAST	Banking Euro Area Stress Testing Model	Major 89 banks (approximately 70 percent of total loans outstanding of financial institutions)	Yes	Budnik et al. [2019], Budnik et al. [2020], ECB [2021b]
(Euro)	STAMP€	Stress-Test Analytics for Macroprudential Purposes in the euro area	Major banks (Approximately 100 banks which differs according to the model)	Yes	Henry and Kok [2013], Dees, Henry, and Martin [2017]
Bundesbank (Germany)		Satellite models	Major 12 banks and small financial institutions	No	Bundesbank [2015]
BdF/ACPR (France)	MERCURE	Modele d'Evaluation des Risques du seCteUR financiEr	Major banks and small financial institutions	No	Camara et al. [2015]
BOE (The UK)	RAMSI	Risk Assessment Model of Systemic Institutions	Major banks (Approximately 10 banks)	One way loop from the economic fluctuation to the bank lending	Aikman et al. [2011], Alessandri et al. [2009], Oliver et al. [2012]
BOC (Canada)	MFRAF	Macrofinancial Risk Assessment Framework	Major banks (D-SIBs)	No	Fique [2017]
(Reference) Bank of Japan	FMM	Financial Macro-econometric Model	109 banks and 247 <i>shinkin</i> banks (approximately 80 to 90 percent of total loans outstanding of depository financial institutions)	Yes	

Appendix 6: Stress Testing Models Developed by the Central Banks in America and Europe

Note: 1. This chart is based on the most recent literature on each model.

2. The supervisory stress test models are used in the stress testing for approval of capital plans for banks in US jurisdictions.